Family Expenditure Survey User Documentation 1996-97

SECTION 5 Database Changes

DATABASE CHANGES Explanatory Notes

This section of the documentation describes the way in the 1996-97 database has changed in relation to the 1995-96 database. Changes are described in terms of the names of the affected variables and codes. This section of the User Documentation is in 13 parts.

Parts 1-3 - Changes to raw variables

Changes to raw variables are listed in terms of those that are new for 1996-97, those that have been deleted from 1995-96 and those which have been amended from 1995-96, including a description of the change.

Parts 3-5 - Changes to derived variables

Many derived variables remain unchanged in name, but it should be remembered that there may well be changes in the contents of the variables that feed into them. You should check the compositions of the variables you intend using. Changes to derived variables are listed in terms of those that are new for 1996-97, those that have been deleted from 1995-96, and those which have been amended from 1995-96, including a description of the change.

Parts 6-8 - FES expenditure codes

Expenditure codes are listed and explained in Section 4 of the User Documentation. Part 6 details expenditure codes which are new to 1996-97, Part 7 details expenditure codes which have been deleted from 1995-96 and Part 8 shows expenditure codes which have amended from 1995-96 - details of the changes are given.

Parts 9-11 - Changes to tables

Part 9 details new tables which have been set up for 1996-97 and describes their contents. Part 10 details tables that have been deleted from 1995-96.

Part 12 - Major changes

This part selects and highlights some of the major changes in definition and data collection coming through from the questionnaires, the diary and the data processing.

Part 13 - Questionnaire changes

This part is a comprehensive list of the changes that have been made to the questionnaire in terms of new questions, deletions and wording and coding changes. It is ordered by 1996-97 question reference number.

Raw Variable Changes for 1996-97

1996-97: SECTION 5, PART 1 New raw variables for 1996-97

New raw variab	les for 1996-97	Family Expenditure Surv	
Variable	Description	Table Name	Question Reference
Household Question	onnaire		
adethnic_ethgrp	ADETHNIC: which group do you belong?	adethnic	H_20.85
agedob	(As you are under 20, may I just check)	adult	H_20.30
bedroom	How many bedrooms are there?	address	H_50.70
busdwell	Address attached to business premises?	address	H_50.25
busroom	Are any of the rooms you have mentioned	address	H_50.75
cagedob	(As you are under 20, may I just check)	child	H_30.30
cdplayer	Do any of you have a Compact Disc player	hhold2	H_190.65
chethnic_ethgrp	CHETHNIC: which group do you belong?	chethnic	H_30.75
clea	Do you make any LEASING payments?	vused	H_290.80
cleaamt	How much was last leasing/hire payment?	vused	H_290.85
computer	Do any of you own a home computer?	hhold2	H_190.70
ctirreg	Please explain when amounts paid and how	hhold1	H_150.55
ctpays	Domestic sewerage charge,Band A Scotland	ctpays	H_10.81
ctwatann	What is the total amnt payable for year?	hhold1	H_150.65
dishwash	Do you have a Dish washer in your hhold?	hhold2	H_190.45
drier	Do you have a tumble drier in your hhold	hhold2	H_190.40
dvclea	WEV of last car leasing/hire payment	vused	H_290.100
dvexp	WEV of expenditure pd from os the hhold.	hhold2	H_650.30
dvhben	WEV of amnt of Housing benefit allowed	rents1	H_70.60
dvhbso	WEV of amnt deducted for other services	rents1	H_70.130
dvhbsv	WEV of Housing Benefit for Services amnt	hbservam	H_70.110
dvhbws	Housing Benefit: water / sewerage amount	rents1	H_70.90
dvrtws	Rent: Amnt incl for water/sewerage.	rents1	H_70.150
dvsv	RENT: amnt incl for services in rent pd.	rents1	H_70.175
dvwsew	WEV of combined water & sewerage rates?	hhold1	H_90.75
exrent	May I check, had you been renting this h	mort	H_110.30
flattype	Is the flat a conversion or purpose blt?	address	H_50.15
freezer	Do you have a deep freezer in yr hhold?	hhold2	H_190.25
fridge	Do you have a refridgerator in yr hhold?	hhold2	H_190.30
ftedchk	May I check, have you been in continuous	adult	H_20.70
furnish	Is the accommodaton provided:	tenure	H_60.25
hbenamt	How much benefit was recvd last time?	rents1	H_70.50
hbenefit	Some people qualify for Housing Benefit,	rents1	H_70.35
hbseramt	How much is the amount on the statement	hbservam	H_70.105
hbserv00	SHOW CARD C1:Any amounts for these items	rents1	H_70.95

Family Expenditure Survey

Yanny Expenditure Survey			Question
Variable	Description	Table Name	Reference
hbserv01	SHOW CARD C1. any of these items?	rents1	H_70.95
hbserv02	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv03	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv04	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv05	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv06	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv07	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv08	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv09	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv10	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbserv11	SHOW CARD C1. Any of these items?	rents1	H_70.95
hbsewer	And on the statement, is there an amount	rents1	H_70.80
hbsoamt	What is the total amount deducted for th	rents1	H_70.125
hbsother	What is are this these services(s)?	rents1	H_70.120
hbstmt	For people on Housing Benefit, the Counc	rents1	H_70.70
hbsvoth	Are there any other services included in	rents1	H_70.115
hbwater	On the document, is there an amount for	rents1	H_70.75
hbwsamt	What is the amount for water/sewerage?	rents1	H_70.85
hhold_acorn	Acorn code - added from sample file	hhold1	H_10.25
hhold_expamt	How much in total was given to you?	hhold2	H_650.20
hhold_exppc	What period did this cover?	hhold2	H_650.25
hhold_exptyp00	Does anyone o/s hhold, give you the money	hhold2	H_650.5
hhold_exptyp01	Does anyone o/s hh give you the money	hhold2	H_650.5
hhold_exptyp02	Does anyone o/s the hh give you the mony	hhold2	H_650.5
hhold_exptyp03	Does anyone o/s the hh give you the mony	hhold2	H_650.5
hhold_exptyp04	Does anyone o/s hhold give you the money	hhold2	H_650.5
hhold_exptyp05	Does anyone o/s hhold give you the money	hhold2	H_650.5
hhold_exptyp06	Does anyone o/s hhold give you the money	hhold2	H_650.5
hhold_exptyp07	Does anyone o/s hhold give you the money	hhold2	H_650.5
hhold_itemtypo	What was the other expenditure?	hhold2	H_650.10
hhold_lacode	Local Authority Code	hhold1	H_10.20
holhotl_how	How did you pay for your holiday hotel?	holhotl	H_490.40
holhotl_lhp	Are you still paying the holiday loan/HP	holhotl	H_490.45
holscat_how	HOLSCAT:Did you pay by loan/HP/credit?	holscat	H_500.40
holscat_lhp	HOLSCAT: Are you still paying the loan?	holscat	H_500.45
hsetype	Is house: detached / semi or terraced?	address	H_50.10

New raw variab	les for 1996-97	Family Expenditure Su		
Variable	Description	Table Name	Question Reference	
inshow	HOLPACK: Did you pay by loan/HP/credit?	holpack	H_480.130	
inslhp	HOLPACK: Are you still paying the loan?	holpack	H_480.135	
intmth	How many months does this cover? IF LESS	mort	H_110.165	
lac96	1996 LOCAL AUTHORITY CODE	hhold1	H_10.100	
living	May I just check, are you living with so	adult	H_20.40	
memhld	Interviewer Code	adult	H_20.45	
microwve	Do you have a microwave oven in yr hhld?	hhold2	H_190.50	
mkrcclea	Lease has been abated?	vused	H_290.90	
mkrcrent	(Rent has been abated)	rents1	H_70.10	
mkrcsew	(Sewamt has been abated)	hhold1	H_90.50	
mkrcwat	(Watamt has been abated)	hhold1	H_90.30	
mkrcwsew	(WSewAmt has been abated)	hhold1	H_90.70	
mobile	Do you have a mobile phone in yr hhold?	hhold2	H_190.60	
onbsroom	How many rooms are used for business?	address	H_50.80	
othactyp	What type of accomodation is it?	address	H_50.20	
othsrv	What is are this these other services?	rents1	H_70.165	
packhow	HOLPACK: Did you pay by loan/HP/credit?	holpack	H_480.55	
packlhp	HOLPACK: Are you still paying the loan?	holpack	H_480.60	
penschk	PENDET:Same Pers Pen as covers mortgage?	pendet	H_160.45	
percelea	VUSED: Period covered by leasing payment	vused	H_290.95	
perchben	How long did this cover?	rents1	H_70.55	
percrent	How long did this cover?	rents1	H_70.15	
percsew	What period did this cover?	hhold1	H_90.45	
percwat	What period did this cover?	hhold1	H_90.25	
percwsew	What period did this cover?	hhold1	H_90.65	
phone	Phone Item number	phone	H_200.20	
phone_anymore	Do any of you pay for any more phones?	phone	H_200.125	
ptbsroom	How many rms are used prtly fr business?	address	H_50.85	
rebate	You said that you paid no rent last time	rents1	H_70.40	
rebate0	Can I just check, what is the reason for	rents1	H_70.45	
rentfrom	Who was it rented from ? PROMPT AS NECE	mort	H_110.35	
rooms	How many rooms do you have altogether in	address	H_50.60	
roomshar	Are any of these rooms shared with anyon	address	H_50.65	
satdish	Do you have a satellite TV dish?	hhold2	H_190.15	
serinc	Does rent include any services?	rents1	H_70.155	

H_70.160

rents1

SHOW CARD C1 Which services?

serinc00

Family Expenditure Survey

Variable	Description	Table Name	Question Reference
	.		
serinc01	SHOW CARD C1 Which services?	rents1	H_70.160
serinc02	SHOW CARD C1 Which services?	rents1	H_70.160
serinc03	SHOW CARD C1 Which services?	rents1	H_70.160
serinc04	SHOW CARD C1 Which services?	rents1	H_70.160
serinc05	SHOW CARD C1 Which services?	rents1	H_70.160
serinc06	SHOW CARD C1 Which services?	rents1	H_70.160
serinc07	SHOW CARD C1 Which services?	rents1	H_70.160
serinc08	SHOW CARD C1 Which services?	rents1	H_70.160
serinc09	SHOW CARD C1 Which services?	rents1	H_70.160
serinc10	SHOW CARD C1 Which services?	rents1	H_70.160
serinc11	SHOW CARD C1 Which services?	rents1	H_70.160
serinc12	SHOW CARD C1 Which services?	rents1	H_70.160
servamt	How much, in total, was included for the	rents1	H_70.170
servnum	Rents\\Services Card C1 Item Number	hbservam	H_70.100
sewerpay	Do you pay sewerage rates or charges? IN	hhold1	H_90.10
sewsep	Do you pay separate or combined water an	hhold1	H_90.15
stord_pabstor0	INTERNAL OFFICE CODE	stord	H_530.25
stord_pabstor1	Code in office	stord	H_530.25
sublet	Do you have formal agrmnt to let/sublet?	address	H_50.50
sublety	Who is that?Close relative/other or non?	address	H_50.55
telephon	Do you have a Telephone?	hhold2	H_190.55
telfilt	The next questions are about telephone a	hhold2	H_200.10
teltype	PHONE: What type of service is this?	phone	H_200.25
tenure_hholde00	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde01	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde02	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde03	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde04	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde05	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde06	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde07	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde08	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde09	In whose name is the property owned or r	tenure	H_60.5
tenure_hholde10	In whose name is the property owned or r	tenure	H_60.5
tvbw	Do you have a BLACK & WHITE TV?	hhold2	H_190.10
tvcol	Do you have a COLOUR TV set?	hhold2	H_190.5

1996-97 : SECTION 5, PART 1 New raw variables for 1996-97

New raw variable	s for 1996-97	Family Exp	enditure Surv
Variable	Description	Table Name	Question Reference
video	Do you have a Video recorder?	hhold2	H_190.20
washmach	Do you have a Washing machine in yr hhld	hhold2	H_190.35
waterpay	Do you pay water rates?	hhold1	H_90.5
worsinc	Were water/sewerage chrgs incd in rent?	rents1	H_70.135
wsewamt	How much did you actually pay last time?	hhold1	H_90.60
wsincamt	How much was included for water/sewer?	rents1	H_70.145
Income questionnair	e		
allow_anymore	Any more allowances in the last 12 month	allow	I_230.55
allow_dvall	WEV of amount of allowance received.	allow	I_230.45
allow_expnum	Expenses paid outsider item number	allow	I_230.10
allow_perwho	Which person gave you the allowance?	allow	I_230.20
atwork3	Have you been at work in last 3 days?	ilo	I_10.95
awayfrom	Do you hve a jb/busnss you are away from	ilo	I_10.46
benr1_bncard00	Are you receiving any of these benefits?	benr1	I_140.10
benr1_bncard01	Are you receiving - child benefit?	benr1	I_140.10
benr1_bncard02	Are you receiving - one parent benefit?	benr1	I_140.10
benr1_bncard03	Are you receiving - Guardian's Allownce?	benr1	I_140.10
benr1_bncard04	Are you receivg - Invalid care allownce?	benr1	I_140.10
benr1_bncard05	Are you receiving - Retirement pension?	benr1	I_140.10
benr1_bncard06	Are you recvng - Old person's pension?	benr1	I_140.10
benr1_bncard07	Are you receiving - a Widow's pension?	benr1	I_140.10
benr1_bncard08	War, Severe disablmnt or disablty alwnce	benr1	I_140.10
benr1_bncard09	None of the benefits shown on card R1!	benr1	I_140.10
benr1amt_amtbenx	What was amount of benefit last recvd?	benr1amt	I_140.15
benr1amt_perbenx	What period did this (amount) cover?	benr1amt	I_140.20
benr2_bncardr0	Are you receiving any of these benefits?	benr2	I_150.10
benr2_bncardr1	Which benefit are you receiving?	benr2	I_150.10
bens1_bncard00	Card S1: Which benefits shown received?	bens1	I_160.15
bens1_bncard05	Card S1: Sickness,Incapacity/invalidity?	bens1	I_160.15
bens2_bncards1	Card S2 - which benefit received?	bens2	I_170.10
bent1_amtbenx	BENT1: Amount of benefit last received?	bent1	I_180.65
bent1_bncard00	Card T3: Which benefit have you received	bent1	I_180.10
bent1_bncard01	Card T1: Which benefit have you recvd?	bent1	I_180.10
bent1_bncard02	Card T1: Which benefit have you recvd?	bent1	I_180.10
bent1_bncard03	Card T1: which benefit have you received	bent1	I_180.10
bent1_bncard04	Card T1: Which benefit have you recvd?	bent1	I_180.10

Family Expenditure Survey

Variable	Description	Table Name	Question Reference
bent1_numwks	Card T1: How many wks was benefit rcvd?	bent1	I_180.60
bent1_perbenx	Card T1: What period did this cover?	bent1	I_180.70
bent1_rcptnow	Card T1: are you recvg this benefit now?	bent1	I_180.75
bkimp	IMPUTED income from BANK account	variousi	I_310.27
bknam	What are the names/types of account held	variousi	I_310.26
bldimp	Building Society IMPUTED income	variousi	I_300.27
bldnam	what are the names of the accounts held?	variousi	I_300.26
bothimp	IMPUTED income from OTHER bank/society	variousi	I_340.26
busaccts	Were annual business accounts prepared?	sejob	I_60.15
cardr1	Prompt Card R1 benefits type Item Number	benr1amt	I_140.14
cclamt	Amount of Community care grant recvd?	bent1	I_180.45
checktax	Were either IT or NI deducted at source?	sejob	I_60.140
chkincom	Is average income net or gross of IT/NI?	sejob	I_60.175
deducts8	Ninth type of deduction from main pay	paymain	I_50.110
dvbkimp	DV for imputed Bank income	variousi	I_310.28
dvbldimp	DV for imputed B.S. income	variousi	I_300.28
dvbothi	DV for other bank income	variousi	I_340.27
dvccg	WEV of Community Care grant, amnt recvd?	bent1	I_180.50
dvfaml	WEV of FAMILY CREDIT LUMP SUM received.	bent1	I_180.20
dvguar	WEV of Guardians allowance amount recvd?	benr1	I_140.27
dvins	WEV for private medical insurance.	paymain	I_50.180
dvnid	WEV of Natl Insurnce deducted last time.	sejob	I_60.170
dvnsimp	WEV of IMPUTED NS Invest acc income	variousi	I_330.27
dvntimp	WEV for IMPUTED Nat Sav ord acc income	variousi	I_320.27
dvonep	WEV of one parent benefit - amount rcvd?	benr1	I_140.26
dvothimp	WEV of imputed share income	shareinc	I_350.50
dvownamt	WEV of drawings for non-business	sejob	I_60.105
dvownot	WEV inc frm business for own use	sejob	I_60.120
dvprbef	WEV of share of profit before tax	sejob	I_60.80
dvprof	WEV for PROFIT made from business.	sejob	I_60.60
dvseinc	Weekly or monthly income from the busns.	sejob	I_60.135
dvseni	WEV of your last NI payment	sejob	I_60.195
dvsenii	How much was amnt of NI lump sum payment	sejob	I_60.225
dvsenil	Total lump sum NI payment in lst 12 mths	sejob	I_60.240
dvsetax	Total tax paid in last 12 months.	sejob	I_60.210
dvtaxd	WEV of income tax deductd last time	sejob	I_60.155

\mathbf{F}	amily	Exper	nditure	Survey
--------------	-------	--------------	---------	--------

Variable	Description	Table Name	Question Reference
dvtesimp	WEV of IMPUTED TESSA income	variousi	I_290.33
famlamt	What was the amnt of Family Credit recvd	bent1	I_180.15
ftpt	Are you working full or part time?	jobmain	I_20.70
funeramt	Amnt of Grant rcvd for funeral expenses?	bent1	I_180.25
hournopa	How many hours UNPAID o/t usually worked	paymain	I_50.250
hourot	How many hours PAID o/t usually worked?	paymain	I_50.245
hourus	How many hours a week do you usu work?	paymain	I_50.240
iemstat1	Imputed employment status	jobcodes	I_370.30
inco	Describe other sources of income.	sejob	I_60.250
indper	Who gave you this payment?	allow	I_230.25
insamt	How much was deducted for priv med ins?	paymain	I_50.175
job2_dirni	Were NI deductions deducted at source?	job2	I_30.31
job2_etype	Describe your employment situation?	job2	I_30.26
job2_indtitle	Enter a short title for industry	job2	I_30.6
job3_dirni	In this job were NI contribs ded at srce	job3	I_40.31
job3_etype	Description of employment situation.	job3	I_40.26
job3_indtitle	Enter a short title for the industry.	job3	I_40.6
jobbus	Do you think you have a job or business?	sejob	I_60.10
jobmain_dirni	Were your NI contribs deductd at source?	jobmain	I_20.36
jobmain_etype	Description of employment.	jobmain	I_20.31
jobmain_indtitle	Short title for industry.	jobmain	I_20.11
nidamt	How much Natnl ins was deducted lst time	sejob	I_60.160
noinc0	No income from self-employment.	sejob	I_60.245
noinc1	No income from self-employment.	sejob	I_60.245
noinc2	No income from self-employment	sejob	I_60.245
noinc3	No income from S-E - other sources?	sejob	I_60.245
nsimp	IMPUTED NS investment income	variousi	I_330.26
ntimp	IMPUTED NS ordinary income.	variousi	I_320.26
oempstat	Original employment status `	jobmain	I_20.75
othimp	EDITOR enter imputed share income	shareinc	I_350.45
othnam	What are the names of the accounts held?	shareinc	I_350.40
ownamt	On average, how much taken each month?	sejob	I_60.100
ownotamt	How much each mnth taken fr personal use	sejob	I_60.115
ownother	Receive any other income from this busns	sejob	I_60.110
ownsum	CARD Q1: Drawings for non-business purps	sejob	I_60.95
partdisp	INTERVIEWER, code 1 to continue.	sejob	I_60.25

Family Expenditure Survey

Variable	Description	Table Name	Question Reference
payinc0	Did take home pay incld any items in P1	paymain	I_50.255
payinc1	Did pay incld any items on card P1	paymain	I_50.255
payinc2	Did pay include any items on card P1?	paymain	I_50.255
payinc3	Did pay include any items on card P1	paymain	I_50.255
pensinc_anymore	PENSION:Any othr incme frm sources on U1	pensinc	I_200.90
percnid	How long did this cover?	sejob	I_60.165
percseni	How long did this cover? (NI payment)	sejob	I_60.190
perctaxd	How long did this cover?	sejob	I_60.150
prbefore	What was yr shre of the profit before tx	sejob	I_60.75
profdocs	What was your share of the profit/loss?	sejob	I_60.45
profit1	Enter the profit/loss amount.	sejob	I_60.50
profit2	Did the answer refer to a profit or loss	sejob	I_60.55
profni	Is the profit befor/after paymnt of NI	sejob	I_60.70
proftax	Is the profit before deduction of tax?	sejob	I_60.65
se1	Most recent period accounts prepared for	sejob	I_60.30
se2	ENTER END OF PERIOD (etc)	sejob	I_60.35
seinc	What is your weekly/monthly income.	sejob	I_60.125
seincwm	Was that weekly or monthly income?	sejob	I_60.130
seniiamt	How much was the NI lump sum payment	sejob	I_60.220
seniinc	Any lump sum of (cl 4)NI based on profit	sejob	I_60.215
senilamt	What was your total lump sum payment	sejob	I_60.235
senilump	Have you paid any lump sum NI contributs	sejob	I_60.230
seniramt	How much was your last Nat Ins payment?	sejob	I_60.185
senireg	Do you pay a regular NI contribution?	sejob	I_60.180
setax	Have you made any other Inc Tax payments	sejob	I_60.200
setaxamt	How much dd you py in lst 12 mths(total)	sejob	I_60.205
seweeks	How many weeks does this cover?	sejob	I_60.40
sole	Working on your own or in partnership?	sejob	I_60.20
taxdamt	How much income tx was deducted lst time	sejob	I_60.145
tesimp	IMPUTED income from TESSA	variousi	I_290.32
tesnam	With which Bank/BS do you have yr TESSA?	variousi	I_290.31
variousi_bncardt4	Benefits revd under hospital savings scm	variousi	I_190.10
waiting	Are you waiting to start job already obt	ilo	I_10.70
whynopro	Why was respo unable to give a prft fig.	sejob	I_60.85
workacc	Separate accounts for work / private?	sejob	I_60.90
workyr	In which yr dd you strt work in this job	jobmain	I_20.65

1996-97 : SECTION 5, PART 2

1995-96 raw variables that have been deleted for 1996-97		Family Expenditure Survey
Variable	Description	Table Name
alltype1	CARD V: second allowance type received	variousi
alltype2	CARD V: third allowance type received	variousi
alltype3	CARD V: fourth allowance type received	variousi
alltype4	CARD V: fifth allowance type received	variousi
atwork	Have you been to work today?	ilo
away3	Have you been away for more than 3 days?	ilo
benr1_amtbenx	Amount of child benefit etc., received	benr1
benr1_amtbenx1	Amount of NI retirement pension received	benr1
benr1_amtbenx2	Amount of NI Widows benefit received	benr1
benr1_bncardr	Prompt card R1: which benefit receiving	benr1
benr1_perbenx	Period covered by Child benefits	benr1
benr1_perbenx1	Period covered by retirement pension	benr1
benr1_perbenx2	Period covered by Widows benefits	benr1
benr2_bncard01	Card R2 - other benefit received?	benr2
benr2_bncard02	Card R2 - other benefit received?	benr2
benr2_bncard03	Card R2 - other benefit received?	benr2
benr2_bncard04	Card R2 - other benefit received?	benr2
benr2_bncard05	Card R2 - other benefit received?	benr2
bens1_bncards1	Show prompt card S1.	bens1
bent1_bncardt	Prompt card T1: which benefit received?	bent1
bent1_bncardt1	Prompt card T1: other benefit received?	bent1
bent1_bncardt2	Prompt card T1: other benefit received?	bent1
bent1_bncardt3	Prompt card T1: other benefit received?	bent1
bent1amt_amtbenx	What amount of benefit was last received	bent1amt
bent1amt_numwks	For how many weeks did you recv benefits	bent1amt
bent1amt_perbenx	What period did this cover?	bent1amt
bent1amt_rcptnow	Are you receiving (T1) benefits now?	bent1amt
bncard06	Card R2 - 8 benefits - any received?	benr2
bncard07	Card R2 - 8 benefits - any received?	benr2
bncardr1	Prompt card R1: other benefit receiving	benr1
bncardr2	Prompt card R1: other benefit receiving	benr1
bncardr3	Prompt card R1: other benefit receiving	benr1
bncards1	One of three benefits shown on Card S2	bens2
bncards2	One of three benfits shown on card S2	bens2

Family Expenditure Survey

1995-96 raw variables that have been deleted for 1996-97

Variable	Description	Table Name
1 1.4	CADD TO COLL CA	
bncardt4	CARD T2: fifth benefit type received	variousi
cardt1	Type of benefit received(prompt card T1)	bent1amt
cccwinrt	Council tax (& water chge) incl in rent? Amount included in rent for council tax	rents 1
ccinrtam		rents1
cper	Period covered by last combined payment	rents2
ctfrq	Period covered cncl tax exemption	hhold
ctfrq	Period covered cncl tax exemption	hhold1
cttrrdam	Amnt transitional rdetn in enel tax pmnt	hhold
cttrrdam	Amnt transitional rdetn in enel tax pmnt	hhold1
cttrrdpc	Period of cncl tax transitional rdctn	hhold
cttrrdpc	Period of cncl tax transitional rdctn	hhold1
cttrred	Transitional rdctn in last cncl tax pmnt	hhold
cttrred	Transitional rdctn in last cncl tax pmnt	hhold1
cwinrtam	Amount in rent for council water charge	rents1
doccon	Documents consulted in transaction	hexp2
docest	SE: profit	sejob
doco	SE: type of other documents consulted	sejob
durable1	Type of second durable - if any	hhold
durable1	Type of second durable - if any	hhold1
durable2	Type of third durable - if any	hhold
durable2	Type of third durable - if any	hhold1
durable3	Type of fourth durable - if any	hhold
durable3	Type of fourth durable - if any	hhold1
durable4	Type of fifth durable - if any	hhold
durable4	Type of fifth durable - if any	hhold1
durable5	Type of sixth durable - if any	hhold
durable5	Type of sixth durable - if any	hhold1
durable6	Type of eighth durable - if any	hhold
durable6	Type of eighth durable - if any	hhold1
durable7	Type of eighth durable - if any	hhold
durable7	Type of eighth durable - if any	hhold1
durable8	Type of ninth durable - if any	hhold
durable8	Type of ninth durable - if any	hhold1
durable9	Type of tenth durable - if any	hhold

1996-97 : SECTION 5, PART 2

1995-96 raw variables that have been deleted for 1996-97		Family Expenditure Survey
Variable	Description	Table Name
durable9	Type of tenth durable - if any	hhold1
durables	Type of first durable - if any	hhold
durables	Type of first durable - if any	hhold1
dvcc	WE of council tax element of rent	rents1
dvct3	WE of council tax transitional relief	hhold
dvct3	WE of council tax transitional relief	hhold1
dvctyear	CT: WEV of ctamt where ctinstal = 1	hhold
dvctyear	CT: WEV of ctamt where ctinstal = 1	hhold1
dvcw	WE of water charge element of rent	rents1
dvhb	WE of services element of rent	rents1
dviwb	Industrial widows benefit, amnt revd	bent1
dvprls	Self-employed profit or loss	sejob
dvrb	WE of last rent rebate	rents1
dvrra	WE of housing benefit allowed in rent	rents1
dvse	WE for sewerage rates	rents2
dvsei	SE: Income tax paid from business.	sejob
dvseit	Self-employment imputed profit	sejob
dvsept	SE: Partners share in the business	sejob
dvser	SE: How much taken out for personal use?	sejob
dvsprt	WE on spare parts in last 3 months	parts
dvsps	Amount received from absent spouse	variousi
dvsva	Value of services in last rent payment	rents1
dvws	WE for water rates	rents2
dvwsa	WE of water	hhold
dvwsa	WE of water	hhold1
dvwsc	WE for combined water	rents2
dvwwb	War widows benefit, amount received	bent1
empdeds	Was the deduction made at source?	pensinc
empmtone	Other pensions from previous employer	pensinc
estopin	SE: interviewers opinion of accuracy	sejob
expdesc	What household expenditure was paid?	hhexp
expg_anymore	Repeat questions	expg
expg_dvexp	WE of money given	expg
expg_expamt	Amount given	expg

Family Expenditure Survey

1995-96 raw variables that have been deleted for 1996-97

Variable	Description	Table Name
expg_expnum	Expenses given item number (index)	expg
expg_exppc	Period covered by amount given	expg
expg_itemtype	Type of expenditure item acquired	expg
expg_itemtypo	Type of housing expnd item acquired	expg
expperc	What period did this cover?	hhexp
filt12	What benefits shwn on card S1 received?	bens1
filtmt	Have you received any Maternity benefits	bens2
filto	Received any benefit shown on card T1?	bent1
filtp	Which benefits (card R2) are you rcvg?	benr2
filtpr	Are you receiving any of these benefits?	benr1
hbserv	Housing benefit including any services	rents1
hbsv0	Type of service included in housing bnft	rents1
hbsv1	Type of service included in housing bnft	rents1
hbsv2	Type of service included in housing bnft	rents1
hbsv3	Type of service included in housing bnft	rents1
hbsv4	Type of service included in housing bnft	rents1
hbsv5	Type of service included in housing bnft	rents1
hbsvo	Type of service included in housing bnft	rents1
hhexp_anymore	Does your (spouse) pay any other expnses	hhexp
hhexp_dvexp	WEV of Total h	hhexp
hhexp_expamt	Total of household expenses paid direct	hhexp
hhexp_expnum	Household expense number (index)	hhexp
hhold_dummy	Dummy variable - not used	hhold
hhold_dummy	Dummy variable - not used	hhold2
hhold_filter21	Expnd items paid by non-employer	hhold
hhold_filter21	Expnd items paid by non-employer	hhold2
hhold_wsamt	Amount of last water	hhold
hhold_wsamt	Amount of last water	hhold2
hhold_wsfrq	Period covered by last water	hhold
hhold_wsfrq	Period covered by last water	hhold2
ina220	How many hours do you work a week	paymain
ina244	How many hours a week overtime worked?	paymain
intperbm	Start month of last 12 month int payment	mort
intperby	Start year of last 12 month int payment	mort

1996-97 : SECTION 5, PART 2

1995-96 raw variables that have been deleted for 1996-97		Family Expenditure Survey
Variable	Description	Table Name
intperem	End month of last 12 month int payment	mort
intperey	End year of last 12 month int payment	mort
lac	Local authority code	hhold
lac	Local authority code	hhold2
lookyt	Were you looking for a place on a GTS?	ilo
malinpay	Motoring allowance in pay, revd lst time	paymain
matstrat	6 weeks at higher or lower rate?	bens2
mkrc070	Abatement marker for last rent payment	rents1
mkrc084	Abatement marker for water rates	hhold
mkrc084	Abatement marker for water rates	hhold2
mkrc084e	Abatement marker for sewerage rates	hhold
mkrc084e	Abatement marker for sewerage rates	hhold2
mkrc085	Abatement marker for water	hhold
mkrc085	Abatement marker for water	hhold2
mkrct3	Abatement marker for spare parts payment	parts
motexp	Refunds for motoring expenses - in pay.	paymain
mrenddtm	SE: Period of self-employment	sejob
mrenddty	SE: period of self-employment	sejob
mrlt1y	SE: How many weeks does this cover?	sejob
mrstadtm	SE: period of self-emplyment	sejob
mrstadty	SELF EMPLOYED dates start period - year	sejob
ninof	Deduction from pay for National Insurnce	paymain
onlybuse	Any rooms used only for business	address
othersv	Other services included in rent payment	rents1
owntype	Purchase method	mort
partbuse	Number of rooms used partly for business	address
parts_vehnum	Vehicle number	parts
parts_vehtype	Spare parts vehicle type	parts
pay2_anticpay	Anticipated pay.	pay2
payef	PAYE deduction for income tax?	paymain
paymain_anticpay	New job - anticipated pay to be given.	paymain
pensinc1	Annuity or personal pension	pension
perc070	Period covered by last rent payment	rents1
perc074	Period covered by last rent payment	rents1

Family Expenditure Survey

1995-96 raw variables that have been deleted for 1996-97

Variable	Description	Table Name
perc465	SE: How often do you withdraw funds?	sejob
perc586	ALLWNCES	variousi
profloss	SE: Was that a profit or a loss?	sejob
rball	Amount last rent rebate	rents1
rbperc	Period covered by last rent rebate	rents1
rent0	100% rent rebate	rents1
rentfree	Property rent free or rent paying	tenure
rentinsv	Last rent payment included any services?	rents1
rentreb	Housing benefit in last rent payment	rents1
rents1_hbamt	Amount for services in last housing bnft	rents1
rents2_wsamt	Amount paid last time for water rates	rents2
rents2_wsfrq	Period covered by last water rates pmnt	rents2
revemp1	Blaise re-coding variable	jobcodes
rmbus	Any rooms used for business	address
rooms_shared	Number of rooms of given type shared	rooms
rooms_type	Type of room	rooms
rooms_used	Number of rooms of given type	rooms
rramt	Amount housing benefit in last rent pmnt	rents1
seenddtm	SE: which 12 months does this cover?	sejob
seenddty	SE: which 12 months does this cover?	sejob
sehours	SE: How many hours do you work per week?	sejob
seimpt	SE: imputation variable	sejob
seinctax	SE: Income tax paid on the business?	sejob
seitxam	SE: How much Income Tax did you pay?	sejob
selt1y	SE: How many weeks does this cover?	sejob
separtnr	SE: Sole owner or partner in business?	sejob
separtsh	SE: How much included for partners share	sejob
sepcom	Separate or combined water	rents2
seprofit	SE: What was the gross profit	sejob
seregs	SE: Drawings (cash) for personal use?	sejob
seregsum	SE: How much do you draw per week	sejob
seshinc	SE: Partners share incldd in profit	sejob
sestadtm	SE: Which period of 12 mths is covered?	sejob
sestadty	SE: Which 12 months does this cover?	sejob

1996-97 : SECTION 5, PART 2

1995-96 raw vai	riables that have been deleted for 1996-97	Family Expenditure Survey
Variable	Description	Table Name
sew	Amount paid last time for sewerage rates	rents2
sewfrq	Period covered by last sewge rates pmnt	hhold
sewfrq	Period covered by last sewge rates pmnt	hhold2
sewper	Period covered by last sewerage payment	rents2
sparpt	Spare parts bought for vehicle last 3mth	hhold
sparpt	Spare parts bought for vehicle last 3mth	hhold2
spoualre	Absent spouse, how much have you recvd?	variousi
spouseal	Absent spouse, allowance received?	variousi
sprtot	Spare parts amount paid in last 3 months	parts
spsaldir	Does you spouse pay any h	variousi
spsaltot	WEV of total h	variousi
sspay	Did your last pay incld Stat Sick Pay?	paymain
sssmpay	Did your last pay incld any SSP or SMP?	paymain
stmore	Spare parts other vehicles last 3 months	parts
stord_pabsto01	Diary code for standing order items	stord
stord_pabstord	Edit variable for coding stord_pabsto1	stord
svamt	Amount for services in last rent payment	rents1
telepres	Telephone in property	hhold
telepres	Telephone in property	hhold2
variousi_dvall	Amount received in allowances, last 12m	variousi
waitng	Are you waiting to start new job?	ilo
watcon	Property connected to mains water	hhold
watcon	Property connected to mains water	hhold2
watfrq	Period covered by last water rates pmnt	hhold
watfrq	Period covered by last water rates pmnt	hhold2
whichdo1	SE: type of documents consulted	sejob
whichdo2	SE: type of documents consulted	sejob
whichdo3	SE: type of documents consulted	sejob
whichdo4	SE: type of documents consulted	sejob
whichdoc	SE: type of documents consulted	sejob
whichsv	Type of service included in rent payment	rents1
whichsv1	Type of service included in rent payment	rents1
whichsv2	Type of service included in rent payment	rents1
whichsv3	Type of service included in rent payment	rents1

Family Expenditure Survey

1995-96 raw variables that have been deleted for 1996-97

Variable	Description	Table Name
whichsv4	Type of service included in rent payment	rents1
whichsv5	Type of service included in rent payment	rents1
wscom	Amnt last time for water	rents2
wscot	Indicator: type of mains water	hhold
wscot	Indicator: type of mains water	hhold2
wssep	Separate or combined water	hhold
wssep	Separate or combined water	hhold2
wssupply	Property connected to mains water	rents2

Date Produced: 3 May 1998

1996-97 : SECTION 5, PART 3

1995-96 raw	variables amended in the 1996-97 dataset		Family Expenditure Survey
Variable	Description	Table Na	nme Decription of Change
Variables that	have moved from one table to another		
allpres	Are you recvg the allowance at present?	allow	Variable moved from table variousi
allrecam	Allowance: How much was last payment?	allow	Variable moved from table variousi
alltype	CARD V: first allowance type received	allow	Variable moved from table variousi
annchg	Annual standing charge on credit cards?	hhold2	Variable moved from table hhold1
bgasamt	Expenditure on bottled gas.	hhold2	Variable moved from table hhold1
bgasbuy	Have you purchased any bottled gas?	hhold2	Variable moved from table hhold1
centh	Do you have central heating, including s	hhold2	Variable moved from table hhold1
chfuel	What fuel does your CH use?	hhold2	Variable moved from table hhold1
chifilt	Paid for Installation of Central Heating	hhold2	Variable moved from table hhold1
chmfilt	Paid for Central Heating Maintenance?	hhold2	Variable moved from table hhold1
creditcd	Do any of you have a credit card?	hhold2	Variable moved from table hhold1
decfilt	House maintenance by contractor?	hhold2	Variable moved from table hhold1
dsperc	What period did this cover?	hhold2	Variable moved from table hhold1
dsschk	May I check, was this the DSS or someone	hhold2	Variable moved from table hhold1
dsselecf	How much did DSS contribute to the last	hhold2	Variable moved from table hhold1
dsselecp	How much did you contribute towards your	hhold2	Variable moved from table hhold1
dssgasf	How much did DSS contribute to the last	hhold2	Variable moved from table hhold1
dssgasp	How much did you contribute towards your	hhold2	Variable moved from table hhold1
dssper	What period did this cover?	hhold2	Variable moved from table hhold1
dvbgas	WEV for cost of Bottled Gas bought	hhold2	Variable moved from table hhold1
dvdssef	How much did DSS cont to 1st ELEC pymnt?	hhold2	Variable moved from table hhold1
dvdssep	What did YOU contrib to last ELEC pymnt?	hhold2	Variable moved from table hhold1
dvdssgf	What did DSS contrib to last GAS paymnt?	hhold2	Variable moved from table hhold1
dvdssgp	What did YOU contrib to last GAS paymnt?	hhold2	Variable moved from table hhold1

Family Expenditure Survey

1995-96 Raw variables amended in the 1996-97 dataset

Variable	Description of Variable	Table Na	me Description of Change
dvdw	Disability working allowance amount revd	benr1	Variable moved from table benr2
dveac	WE of electricity, accnt & other method	hhold2	Variable moved from table hhold1
dvebb	WE of electricity, Board Budgeting Schm	hhold2	Variable moved from table hhold1
dvechg	WE of electricity charge, BBS	hhold2	Variable moved from table hhold1
dverb	WE of electricity rebate, slot meter	hhold2	Variable moved from table hhold1
dvfur	WE of furniture in last three months	hhold2	Variable moved from table hhold1
dvgac	WE of gas, account & other method	hhold2	Variable moved from table hhold1
dvgbb	WE of gas, Board Budgeting Scheme	hhold2	Variable moved from table hhold1
dvgch	WE of gas charge, Board Budgeting Scheme	hhold2	Variable moved from table hhold1
dvgrb	WE of gas rebate, slot meter	hhold2	Variable moved from table hhold1
dvica	Invld care allwnc, amnt rcvd last 12m	benr1	Variable moved from table benr2
dvind	Industrial inj bnft, amnt revd last 12m	bens1	Variable moved from table benr2
dvmatg	Social Fund grant for maternity, amount.	bent1	Variable moved from bens2
dvoil	WE of oil for central heating	hhold2	Variable moved from table hhold1
dvrug	WE of carpets	hhold2	Variable moved from table hhold1
dvsvs	Severe disbl benefit, amnt rcvd (12mths)	benr1	Variable moved from table benr2
dvtela	WEV of telephone, accnt & other method	phone	Variable moved from hhold1
dvtelb	WE of telephone, Budgeting Scheme	phone	Variable moved from hhold1
dvtelc	WE of telephone charge, Budgeting Scheme	phone	Variable moved from hhold1
dvtels	WE of phone, accnt share, non-empl conts	phone	Variable moved from hhold1
dvtvl	WE of TV licence fee paid last 12 months	hhold2	Variable moved from hhold1
dvwar	War disablement pension, amount recvd	benr1	Variable moved from table benr2
eacamt	How much did you pay last time, excludin	hhold2	Variable moved from hhold1
eacper	What period did this cover?	hhold2	Variable moved from hhold1
ebbsamt	How much was your last budgeting scheme	hhold2	Variable moved from hhold1
ebbsper	What period did this cover?	hhold2	Variable moved from hhold1
echgamt	What was the CHARGE on yr lst electric?	hhold2	Variable moved from hhold1

Date Produced: 3 May 1998

1996-97: SECTION 5, PART 3

1995-96 raw variables amended in the 1996-97 dataset

Family Expenditure Survey

Variable	Description	Table Na	me Decription of Change
echgper	What period did this cover?	hhold2	Variable moved from hhold1
elecpay	E2:By which of these methods did you pay	hhold2	Variable moved from hhold1
elecpayo	How do you pay?	hhold2	Variable moved from hhold1
elecrebt	Have you recvd a rebate on your bill?	hhold2	Variable moved from hhold1
erbtamt	How much was the rebate before any deduc	hhold2	Variable moved from hhold1
erbtper	What period did this cover?	hhold2	Variable moved from hhold1
feefilt	CARD I: Have you pd any property fees?	hhold2	Variable moved from hhold1
filtsr	Have any of you pd fr a vehicle service?	hhold2	Variable moved from hhold1
furnamt	Wht in total did you py fr the furniture	hhold2	Variable moved from hhold1
furnfilt	K1: Have you bought any furniture?	hhold2	Variable moved from hhold1
furnhow	Did you pay by	hhold2	Variable moved from hhold1
furnlhp	Are you still paying instalments on the	hhold2	Variable moved from hhold1
gacamt	How much did you pay last time, excludin	hhold2	Variable moved from hhold1
gacper	What period did this cover?	hhold2	Variable moved from hhold1
gaspay	By what method did you pay for your gas?	hhold2	Variable moved from hhold1
gaspayo	How do you pay?	hhold2	Variable moved from hhold1
gasrebt	Have you rcvd a gas rebate in lst 3 mths	hhold2	Variable moved from hhold1
gassupp	Do you have gas supplied to your (part o	hhold2	Variable moved from hhold1
gbbsamt	How much did you pay last time?	hhold2	Variable moved from hhold1
gbbsper	What period did this cover?	hhold2	Variable moved from hhold1
gchgamt	How much was the CHARGE on your last adv	hhold2	Variable moved from hhold1
gchgper	What period did this cover?	hhold2	Variable moved from hhold1
givenby	Who gave you the money for this expendit	hhold2	Variable moved from table expg
grbtamt	How much was the last rebate before any	hhold2	Variable moved from hhold1
grbtper	What period did this cover?	hhold2	Variable moved from hhold1
hhold_address	Address Number	hhold1	Variable moved from hhold2
hhold_area	Area Number	hhold1	Variable moved from hhold2
hhold_filter00	Insce prems on structure/contents/other	hhold1	Variable moved from hhold2
hhold_filter01	Contributions to a Personal Pension schm	hhold1	Variable moved from hhold2
hhold_filter02	Premiums on life/death/endowmnt policies	hhold1	Variable moved from hhold2
hhold_filter03	CARD E1: Any other insurance premiums?	hhold1	Variable moved from hhold2
hhold_filter04	Do you make any pymnts shown on card F1?	hhold2	Variable moved from hhold
hhold_filter05	Have any of you received any refunds?	hhold2	Variable moved from hhold2
hhold_filter06	Do any of you have a season ticket?	hhold2	Variable moved from hhold

Family Expenditure Survey

1995-96 Raw variables amended in the 1996-97 dataset

Variable	Description of Variable	Table Na	me Description of Change
hhold_filter08	Do any of you have a bank account?	hhold2	Variable moved from hhold
hhold_filter09	Payments by standing order/direct debit?	hhold2	Variable moved from hhold
hhold_filter10	FREE welfare milk received?	hhold2	Variable moved from hhold
hhold_filter11	FREE SCHOOL MILK received by children?	hhold2	Variable moved from hhold
hhold_filter12	Have any children had any SCHOOL MEALS?	hhold2	Variable moved from hhold
hhold_filter13	Childrens travel to/from school	hhold2	Variable moved from hhold
hhold_filter14	Do you receive an education grant?	hhold2	Variable moved from hhold
hhold_filter15	Student Loan rcvd fr attending a course?	hhold2	Variable moved from hhold
hhold_filter16	School fees paid in last 3 months?	hhold2	Variable moved from hhold
hhold_filter17	Leisure class fees in last 3 months?	hhold2	Variable moved from hhold
hhold_filter18	Education fees: non-household members?	hhold2	Variable moved from hhold
hhold_filter19	Do you pay maintenance/separation alwnc?	hhold2	Variable moved from hhold
hhold_filter20	Any items refunded by employer (3 mnths)	hhold2	Variable moved from hhold
hhold_filter22	Items paid from outside the household?	hhold2	Variable moved from hhold
hhold_hbamt	HHOLD1: HBen deducted frm lst rent pymnt	hhold1	Variable moved from hhold2
hhold_hhold	Household Number	hhold1	Variable moved from hhold2
hhold_homechks	SSD system variable	hhold1	Variable moved from hhold2
hhold_intdate	Interview Date	hhold1	Variable moved from hhold2
hhold_ratesinc	Does your rent also include rates?	hhold1	Variable moved from hhold2
hhold_spnum0	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum1	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum2	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum3	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum4	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum5	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum6	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum7	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum8	SSD system variable	hhold1	Variable moved from hhold2
hhold_spnum9	SSD system variable	hhold1	Variable moved from hhold2
hhold_where0	Central Heating Installation	hhold2	Variable moved from table hhold
hhold_where1	CENTRAL HEATING MAINTENANCE	hhold2	Variable moved from table hhold
hhold_where2	IMPROVEMETNS AND EXTENSIONS	hhold2	Variable moved from table hhold
hhold_where3	House maintenance: Main or second home?	hhold2	Variable moved from table hhold
impfilt	SHOW PROMPT CARD H1	hhold2	Variable moved from table hhold

1996-97: SECTION 5, PART 3

1995-96 raw variables amended in the 1996-97 dataset

Family Expenditure Survey

Variable	Description	Table Na	me Decription of Change
intstime	(interview start time)	hhold1	Variable moved from hhold2
matgramt	What was the amount of grant received?	bent1	Variable moved from table bens2
mkrc190	(GBBSAmt has been abated)	hhold2	Variable moved from table hhold
mkrc191a	(GChgAmt has been abated)	hhold2	Variable moved from table hhold
mkrc192	(GAcAmt has been abated)	hhold2	Variable moved from table hhold
mkrc201a	(EChgAmt has been abated)	hhold2	Variable moved from table hhold
mkrc202	(EAcAmt has been abated)	hhold2	Variable moved from table hhold
mkrc261a	TelShAmt has been abated.	phone	Variable moved from table hhold
mkrc262	Abatement marker for telephone account	phone	Variable moved from table hhold2
mkrc263	Abatement marker for phone budget scheme	phone	Variable moved from table hhold2
mkrc264	Abatement marker for phone charge acct	phone	Variable moved from table hhold2
mkrc300	(EBBSAmt has been abated)	hhold2	Variable moved from table hhold
mkrcct	(CTAmt has been abated)	hhold1	Variable moved from table hhold
mkrcpy	Has PayRamt been abated?	hhold1	Variable moved from table hhold2
mkrcwt	(CTWat has been abated)	hhold1	Variable moved from table hhold2
mntwho	Who is the maintenance for?	allow	Variable moved from table variousi
movefilt	Payment for storage/etc of furniture etc	hhold2	Variable moved from table hhold
nirate	Domestic rates poundage - N.Ireland only	hhold1	Variable moved from table hhold2
nounits	(number of benefit units)	hhold1	Variable moved from table hhold2
nrv	Net rateable value - Northern Ireland	hhold1	Variable moved from table hhold2
numchil	How many children aged under 16 are ther	hhold1	Variable moved from table hhold2
oilamt	What was the total expenditure on oil in	hhold2	Variable moved from table hhold
oilbuy	Have you purchased any oil for the centr	hhold2	Variable moved from table hhold
othpay	Accomodation charges: Do you pay any?	hhold1	Variable moved from table hhold2
othpay1	Do you pay any of these charges?	hhold1	Variable moved from table hhold2
othpay2	Do you pay any of these charges?	hhold1	Variable moved from table hhold2
othtype	Please specify the other type of charges	hhold1	Variable moved from table hhold2
othwk	Have any of you had any other work done?	hhold2	Variable moved from table hhold
pabnum	EDITOR: Enter your PAB identification nu	hhold1	Variable moved from table hhold2
pafreg	Postal address file region	hhold1	Variable moved from table hhold2
payramt	How much did you pay last time?	hhold1	Variable moved from table hhold2
payrates	Do you pay rates on this accommodation?	hhold1	Variable moved from table hhold2
payrfrq	What period did this cover?	hhold1	Variable moved from table hhold2
perc584	What period does it cover?	allow	Variable moved from table variousi

Family Expenditure Survey

1995-96 Raw variables amended in the 1996-97 dataset

Variable	Description of Variable	Table Nar	me Description of Change
rebtch	May I check, did your rebate cover	hhold1	Variable moved from table hhold2
rugamt	Expenditure on rugs and carpets?	hhold2	Variable moved from table hhold
rugfilt	Have you bought any carpets/rugs?	hhold2	Variable moved from table hhold
rughow	Did you pay by	hhold2	Variable moved from table hhold
ruglhp	Are you still paying instalments on the	hhold2	Variable moved from table hhold
samth	Sample month	hhold1	Variable moved from table hhold2
saqter	Sample quarter	hhold1	Variable moved from table hhold2
scndhome	Second dwelling?	hhold2	Variable moved from table hhold
servd	The date of the LAST vehicle service?	hhold2	Variable moved from table hhold
sewamt	How much did you actually pay last time	hhold1	Variable moved from table hhold2
shinuk	Is the second dwelling in England, Scotl	hhold2	Variable moved from table hhold
showben	SSD system variable	hhold1	Variable moved from table hhold2
survyr	Year of survey	hhold1	Variable moved from table hhold2
telacamt	PHONE ACCOUNT: How much paid last time?	phone	Variable moved from table hhold2
telacper	PHONE ACCNT: What period did this cover	phone	Variable moved from table hhold2
telbgamt	PHONE BUDGET: How much paid last time?	phone	Variable moved from table hhold2
telbgper	PHONE BUDGET: period covered by payment	phone	Variable moved from table hhold2
telchamt	PHONE: Amount of CHARGE on last advice	phone	Variable moved from table hhold2
telchper	PHONE CHARGE: What period was covered?	phone	Variable moved from table hhold
telehow	PHONE: by which method was this paid?	phone	Variable moved from table hhold2
telehowo	PHONE: How do you pay?	phone	Variable moved from table hhold2
telopay	PHONE: Did anyone o/s hhold contribute?	phone	Variable moved from table hhold2
telp	Do you pay anything towards a telephone?	hhold2	Variable moved from table hhold
telshamt	PHONE: How much was your share? (bill)	phone	Variable moved from table hhold2
telshper	PHONE: What period does this cover?	phone	Variable moved from table hhold2
tvlice	How much have you paid for a TV licence	hhold2	Variable moved from table hhold
vintdate	(security copy)	hhold1	Variable moved from table hhold2
watamt	How much did you actually pay for water	hhold1	Variable moved from table hhold2
wsinc	Were water or sewerage charges (rates) i	rents1	Variable moved from table rents2

Banding entry changes

ltresyrs How many years have you lived at this ad address New banding entries

1996-97: SECTION 5, PART 3

1995-96 raw variables amended in the 1996-97 dataset		Family Expenditure Survey		
Variable	Description	Table Nai	me Decription of Change	
season_pabseas1 soccls1	Diary code for season ticket types Social Class	season jobcodes	Banding entries no longer exist New banding entries	
Changes to Datat	ype			
bonam itemdea_litempur	Amount of bonus received Description of item acquired with loan	bonus itemdea	Is now integer4 was integer2 Is now varchar(50) was varchar(15)	
job2_industry	What does your company do/make?	job2	Is now varchar(80) was varchar(40)	
job2_jobtitle	What is your job title?	job2	Is now varchar(30) was varchar(40)	
job3_industry	What does/did the organisation do/make?	job3	Is now varchar(80) was varchar(40)	
job3_jobtitle	What is/was your job title?	job3	Is now varchar(30) was varchar(40)	
jobinc	What kind of job or income?	childinc	Is now varchar(60) was varchar(40)	
jobmain_industry	What does/did the organisation do/make?	jobmain	Is now varchar(80) was varchar(40)	
jobmain_jobtitle	What is/was your job title?	jobmain	Is now varchar(30) was varchar(40)	
linsoth	Type of policy - undefined in linstype	life	Is now varchar(30) was varchar(40)	
minsoth	nsurance policy - undefined in minstype	medins	Is now varchar(30) was varchar(40)	
potyp	What was the purpose of this deduction	pensinc	Is now varchar(50) was varchar(40)	
seg1 soccls1 typefirm	Socio-Economic Group Social Class Firm type from which item/service obtd	jobcodes jobcodes agree2	Is now float4 was integer1 Is now float4 was integer1 Is now varchar(30) was varchar(40)	

Derived
Variable
Changes
for 1996-97

1996-97 : SECTION 5, PART 4

New derived variables for 1996-97

Family Expenditure Survey

Variable	Description	Table Name
a0061	Ethnic origin of people in hh (ad & ch)	set41
a0421	Number of children age under 16	set2
a0422	Number of children age between 7 and 15.	set2
a0423	Number of children completing a diary.	set2
a1331	mortgage - year current mortgage began	set12
a1641	Satellite dish in household	set1
a1651	Mobile phone in household (exc. car phon	set1
a1661	Home computer in household	set1
a2444	Hours unpaid overtime usually worked	set43
b055p	Council sewerage tax - basic wkly amount	set12
b055u	Coun sewerage tax - basic wkly am payble	setunanonh
b056p	Council sewerage tax - weekly amnt paid	set12
b056u	Council sewerage tax - wkly amount paid	setunanonh
b057p	Council sewerage chge after status dscnt	set12
b057u	Council sewerage chge after status dscnt	setunanonh
b1651	Mobile phone budget a	set15
b1661	Mobile telephone account	set15
b1671	Mobile phone budget a	set15
b194a	Rent for TV/Satellite/VCR wkly amt	set20
b194b	Satellite subscription wkly amt	set20
b194c	Cable subscription wkly amt	set20
ctspcnt	Council sewerage chge:amt as % amnt due	setctp
ctsproxy	Council Tax Sewerage Charges (Scotland)	setctp
d030512	Meat uncooked: unspecified	set18
d030512c	Meat uncooked: unspecified	set18c
d0305121	Meat - n.o.specified -large supermarkets	set27a
d030512t	Meat uncooked: unspecified	set18t
d0401011	Beer and lager - large supermarkets	set27a
d0401021	Cider - large supermarkets	set27a
d0401031	Table wines/wines nos - lrge sprmrkts	set27a
d0401041	Champagne/spkling wines - lg sprmrkts	set27a
d0401051	Fortified wines - large supermarkets	set27a
d040106l	Spirits, 1 iquers - large supermarkets	set27a
10401071	Alc sft drinks/alcopops - lg supermrkets	set27a
10401081	Off-licence Alcohol nos - lg spermrkets	set27a
1040108	Alcohol from off-licences nos	set19
d040108c	Alcohol from off-licence nos	set19c
d0401081	Off-licence Alcohol nos-lge supermarkets	set27a
d040108t	Alcohol from off-licence nos	set19t
d040208	Alcohol n.o.s. consumed on licenced prem	set19

Family Expenditure Survey

New derived variables for 1996-97

Variable	Description	Table Name
d040208c	Alcohol n.o.s. consumed on licenced prem	set19c
d040208t	Alcohol n.o.s. consumed on licensed prem	set19t
d0501011	Cigarettes - large supermarkets	set27a
d0501021	Pipe tabacco/tbacco nos-lge supermarkets	set27a
d0501031	Cigars - large supermarkets	set27a
d060101x	Mens outerwear- selected clothing chains	set27a
d060101y	Mens outerwear - large supermarkets	set27a
d060101z	Mens outerwear - charity shops	set27a
d060102x	Womens outerwear-selectd clothing chains	set27a
d060102y	Womens outerwear - large supermarkets	set27a
d060102z	Womens outerwear - charity shops	set27a
d060103x	Boys outerwear- selected clothing chains	set27a
d060103y	Boys outerwear - large supermarkets	set27a
d060103z	Boys outerwear(aged 5-15)- charity shops	set27a
d060104x	Girls outerwear-selected clothing chains	set27a
d060104y	Girls outerwear - large supermarkets	set27a
d060104z	Girls outerwear(aged 5-15)-charity shops	set27a
d060105x	Infants outerwear-selctd clothing chains	set27a
d060105y	Infants outerwear - large supermarkets	set27a
d060105z	Infants outerwear(under 5)-charity shops	set27a
d060201x	Mens underwear- selected clothing chains	set27a
d060201y	Mens underwear - large supermarkets	set27a
d060201z	Mens underwear - charity shops	set27a
d060202x	Womens underwear-selectd clothing chains	set27a
d060202y	Womens underwear - large supermarkets	set27a
d060202z	Womens underwear - charity shops	set27a
d060203x	Childrens underwear-selctd clothg chains	set27a
d060203y	Childrens underwear - large supermarkets	set27a
d060203z	Childrens underwear - charity shops	set27a
d060301x	Mens accessories-selectd clothing chains	set27a
d060301y	Mens accessories - large supermarkets	set27a
d060301z	Mens accessories - charity shops	set27a
d060302x	Womens accessories-selectd clothg chains	set27a
d060302y	Womens accessories - large supermarkets	set27a
d060302z	Womens accessories - charity shops	set27a
d060303x	Childrens accessories-selctd clthg chain	set27a
d060303y	Childres accessories- large supermarkets	set27a
d060303z	Childrens accessories - charity shops	set27a
d060401x	Mens footwear - selected clothing chains	set27a
d060401y	Mens footwear - large supermarkets	set27a

1996-97 : SECTION 5, PART 4

New derived variables for 1996-97

Family Expenditure Survey

Variable	Description	Table Name	
d060401z	Mens footwear - charity shops	set27a	
d060402x	Womens footwear-selected clothing chains	set27a	
d060402y	Womens footwear - large supermarkets	set27a	
d060402z	Womens footwear - charity shops	set27a	
d060403x	Childrens/Infnts ftwear-slctd cltg chain	set27a	
d060403y	Childrens/Infnts ftwear-lge supermarkets	set27a	
d060403z	Childrens/Infants ftwear - charity shops	set27a	
d060404x	Footwear(n.o.s) - selctd clothing chains	set27a	
d060404y	Footwear(n.o.s) - large supermarkets	set27a	
d060404z	Footwear(n.o.s) - charity shops	set27a	
d060501x	Haberdashery - selected clothing chains	set27a	
d060501y	Haberdashery - large supermarkets	set27a	
d060501z	Haberdashery - charity shops	set27a	
d060502x	Clothg mterials(nos)-selctd clothg chain	set27a	
d060502y	Clothg mterials(nos)-large supermarkets	set27a	
d060502z	Clothing materials (nos) - charity shops	set27a	
d0702091	Electric consummables - lrge spermrkts	set27a	
d0703061	Matches - large supermarkets	set27a	
d0704011	Detergents etc large supermarkets	set27a	
d0704021	Disinfectants etc large supermarkets	set27a	
d0705011	Toilet paper - large supermarkets	set27a	
d0706011	Petfood - large supermarkets	set27a	
d0706021	Pet purchase etclargesupermarkets	set27a	
d0802011	Postage/poundage - large supermarkets	set27a	
d0901011	Toilet soap - large supermarkets	set27a	
d0901021	Toiletries - large supermarkets	set27a	
d0901031	Toilet requisites - large supermarkets	set27a	
d0901051	Cosmetics & related - large supermarkets	set27a	
d1003011	Petrol - large supermarkets	set27a	
d1003021	Diesel oil - large supermarkets	set27a	
d1003031	Other motor oils - large supermarkets	set27a	
d110104	Accessories & repairs of bicycles	set14	
d110104c	Accessories & repairs of bicycles etc.	set14c	
d110104t	Accessories & repairs of bicycles etc	set14t	
d1203011	Newspapers - large supermarkets	set27a	
d120302l	Magazines/periodicals-large supermarkets	set27a	
d120304l	Personal stationery - large supermarkets	set27a	
d130206	Cable TV subscription (diary)	set900	
d130206c	Cable TV subscription	set20c	
d130508	National Lottery Saturday draw stakes	set20	

Family Expenditure Survey

New derived variables for 1996-97

Variable	Description	Table Name	
d130508c	National Lottery Saturday Draw Stakes	set20c	
d130508t	National Lottery Saturday draw stakes	set20t	
d130509	National Lottery mid-week draw stakes	set20	
d130509c	National Lottery mid-week draw stakes	set20c	
d130509t	National Lottery mid-week draw stakes	set20t	
d130510	National Lottery draw stakes; both draws	set20	
d130510c	National Lottery draw stakes; both draws	set20c	
d130510t	National Lottery draw stakes; both draws	set20t	
d130608	National Lottery Saturday draw winnings	set20	
d130608c	National Lottery Saturday draw winnings	set20c	
d130608t	National Lottery Saturday draw winnings	set20t	
d130609	National Lottery mid-week draw winnings	set20	
d130609c	National Lottery mid-week draw winnings	set20c	
d130609t	National Lottery mid-week draw winnings	set20t	
d130610	National Lottery draw winngs; both draws	set20	
d130610c	National Lottery draw winngs; both draws	set20c	
d130610t	National Lottery draw winngs; both draws	set20t	
d140106	Contributions to widows & orphans funds	set24	
d140106t	Contributions to widows & orphan funds	set24t	
d140307	Residential care fees for sick & elderly	set24	
d140307c	Residential care fees for sick & elderly	set24c	
d140307t	Residential care fees for sick & elderly	set24t	
p076	SE NI contributions paid last 12 months	set44	
t1143	Other non motorised vehicles,repairs &c	set26	
t1335	Cable TV connection and subscription	set26	
t159	Windfall receipts from gambling etc	set26	
t3112	Offal and other uncooked meats	set26	
t44off	Other alcoholic drinks bght at off licns	set26	
t44on	Other alcoholic drnks bt on licned prems	set26	

1996-97: SECTION 5, PART 5
Family Expenditure Survey 1995-96 derived variables that have been deleted for 1996-97

Variable	Description	Table Name
a122	Rent-not paid-100% rebate or paid direct	set12
a133	Mortgage - years has run	set12
a166	Freezer in household	seti
a2191	Main job se employment period no of wks	set43
a2192	ist subs job se emply period no of wks	set43
a2193	2nd subs job se emply period no of wks	set43
acorn	Acorn geo-demographic code for location	set130
b034p	Coun tax - transitional relief wkly amt	set12
b034u	Coun tax - transitional relief wkly amnt	setunanonh
b251	Car or van - spare parts amount paid	set [4
b3131	Main job SE income - period under 1 yr	set43
b3132	1st subs job SE inc - period under 1yr	set43
b 3133	2nd subs job SE inc - period under lyr	set43
b385	Allowances from absent spouse - amount	set46
cttrpcnt	Council tax trans rel as % of amt due	setctp
d031111	Foods other and undefined	set18
d031111c	Foods other and undefined	set18c
d0311111	Foods other and undefined - large shops	set27a
d031111t	Foods other and undefined	set18t
d0402011	Beer & lager on premises - large shops	set27a
d0402021	Cider on premises - large shops	set27a
d0402031	Table wines on premises -large shops	set27a
d0402041	Champagne on premises-large shops	set27a
d0402051	Fortified wines on premises-large shops	set27a
d040206l	Spirits, liquers on premises-large shops	set27a
d040207i	Alcohol consumed on licensed prems-lg sh	set27a
d0601011	Mens outerwear - large shops	set27a
d0601021	Womens outerwear - large shops	set27a
d0601031	Boys outerwear -large shops	set27a
d0601041	Girls outerwear - large shops	set27a
d0601051	Infants underwear - large shops	set27a
d0602011	Mens underwear - large shops	set27a
d0602021	Womens underwear - large shops	set27a
d0602031	Boys underwear - large shops	set27a
d0603011	Mens accessories - large shops	set27a
d0603021	Womens accessories - large shops	set27a

1995-96: SECTION 5, PART 5
1994-95 derived variables that have been deleted for 1995-96 Family Expenditure Survey

Variable	Description	Table Name
d0603031	Childrens accessories - large shops	set27a
d0604011	Mens footwear - large shops	set27a
d060402I	Womens footwear - large shops	set27a
d0604031	Childrens footwear - large shops	set27a
d0604041	Footwear n o s - large shops	set27a
d0605011	Haberdashery - large shops	set27a
d0605021	Clothing materials - large shops	set27a
d0701011	Furn inc beds and mattresses - Ige shop	set27a
d0701021	Soft floor coverings - large shops	set27a
d0701031	Hard floor coverings - large shops	set27a
d070104l	Bedding exc beds and mattresses - lg sh	set27a
d0701051	Curtains, cushions, etc - large shops	set27a
d0702011	Electric and combi cookers - large shops	set27a
d070202I	Elec wash mach's, spin dryers - lg sho	set27a
d0702031	Elec refrigerators, freezers - lge shop	set27a
d0702041	Dishwashers, microwaves, etc - large shop	set27a
d0702051	Electrical tools - large shops	set27a
d070206l	Minor elec equipment - large shops	set27a
d0702071	Gas cookers - large shops	set27a
d070208I	Other gas appliances - large shops	set27a
d0702101	Repairs to gas and elec - large shops	set27a
d0703011	Kitchen utensils and equipment - lg shop	set27a
d0703021	Kitchen disposables - large shops	set27a
d0703031	China, glass, pottery etc - large shops	set27a
d0703041	Decorative goods - large shops	set27a
d070305l	Other household hardware - large shops	set27a
d0707011	Garden equipment - large shops	set27a
d0707021	Garden tools and accessories - Ige shops	set27a
d0707031	Plants, seeds, etc - large shops	set27a
d0708011	Household goods nos - large shops	set27a
d080206t	Mobile phone account payments	- set15t
d130204t	Satellite tv subscription to channels	set20t
t158	Windfall receipts from gambling etc	set26

Change to banding for frame14

Family Expenditure Survey 1995-96 Derived			es amended in the 1996-97 datase
		Table	
Variable	Description of Variable	Name	Description of Change
Variables	that have moved from one table to another		
d080203c	Telephone Account	set15c	Variable moved from table set900
d080206	Mobile phone account payments	set900	Variable moved from table set15
d100202	Car or Van spare parts	set14	Variable moved from table set900
d130204	Satellite TV subs to channels (diary)	set900	Variable moved from table set20
Banding e	ntry changes		
al31	Hoh-no of months\\years at address	set8	New banding entries
a002	Relationship to head of household	set41	Change to banding entry for frame14
a108	Washing machine in household	set l	Change to banding entry for a108
al 19	Mortgage - source	set12	Change to banding entry for al 19
a128	Gas - method of payment	set13	Change to banding entry for gaspay
a130	Electricity - method of payment	set13	Change to banding entry for elecpay
a141	Possession of video recorder	set1	Change to banding entry for a141
a164	Fridge-freezer or deep freezer in hhold	set1	Change to banding entry for a164
a165	Fridge in household	set l	Change to banding entry for a165
a167	Tumble dryer in household	set l	Change to banding entry for a167
a168	Microwave oven in household	setl	Change to banding entry for a168
a169	Dishwasher in household	set l	Change to banding entry for a169
a170	Compact disc player in household	set1	Change to banding entry for a170
al7l	Possession of television	set l	Change to banding entry for a171
a207	Reason away from work	set43	Change to banding entry for ina207
a209	Pay received while away from work - type	set43	Change to banding entry for ina209
a246	Type of gvt training scheme attended	set50	Change to banding entry for progtype
crcode	Instlmnt credit\\club-item type acquired	set87	Chnge to banding for expend_keytext
ditemcode	Diary expenditure item code	set i 14	Chnge to banding for expend_keytext
hpitem	HP - item acquired	set88	Chage to banding for expend_keytext
litem	Instlmnt credit\\loan-item acquired	set86	Chage to banding for expend_keytext
loanno	Instalment credit\\loan-item no	set86	Chnge to banding for loandea_org
pdrcode	Expenditure item code -adult only	set89	Change to banding for expend_keytext
pdrcodec	Expenditure item code - child only	set89c	Chage to banding for expend_keytext
		_	

setceh

Sex of head of household

sexhoh

1770 70	995-96 Derived variables amended in the 1996-97 dataset		Family Expenditure Surve	
Variable	Description of Variable	Name	Description of Change	
Variables	that have been amended			
a002	Relationship to head of household	set41	Formula and conditions changed	
a006	Marital status	set41	Uses "living & memhld" in derivation	
a010	Age completed continuous ft education	set4 l	Description of raw variable changed	
a095	Social class of head of household	set8	"soccls1" in 7 new formula conditions	
a102	Telephone in household	set l	Now uses raw variable "telephon"	
a108	Washing machine in household	s e t l	Now uses raw variable "washmach"	
alll	Rooms used solely by household	set l	Variables/table & formula changed	
a112	Rooms shared by household	set l	Now uses raw variable "roomshar"	
a116	Category of dwelling	set12	3 new variables & new formulae	
al 17	No of new cars/vans bght lst 12 mths	set l	Period covered extended to 12 months	
al 18	No of s-hand cars/vans bght lst 12 mths	set l	Period covered extended to 12 months	
a121	Tenure type	set12	Variables/table and notes changed	
a131	Hoh-no of months/years at address	set8	New 16 permutation coding frame	
a141	Possession of video recorder	set l	New raw variable "video" used	
a162	No of motorcycles owned or used	set I 4	Two new formulae used in calculation	
a164	Fridge-freezer or deep freezer in hhold	set1	New raw variable "freezer" used	
a165	Fridge in household	set l	New raw variable "fridge" used	
a167	Tumble dryer in household	set l	New raw variable "drier" used	
a168	Microwave oven in household	set l	New raw variable "microwve" used	
a169	Dishwasher in household	set 1	New raw variable "dishwash" used	
a170	Compact disc player in household	set l	New raw variable "cdplayer" used	
al7l	Possession of television	set l	Raw variables "tvbw & tvcol" used	
a200	Employment status	set41	Changes to variables & formulae used	
a204	Government training programme	set4 l	Business start-up scheme not included	
a215	Socio-economic group - main occptn	set43	Major revisions to formula conditions	
a220	Usual wkly hours (exc brks & ovrtme)	set43	New raw variable "hourus" used	
a222	Unemployment buft no of weeks rovd	set48	Condition (cards1) changed (now = 1)	
a223	Unemployment bnft - rcvg at present	set48	Condition (cards1) changed (now = 1	
a228	Income support - no of weeks received	set5l	Condition (cards1) changed (now = 2	
a229	Income support - receiving at present	set51	Condition (cards1) changed (now = 2	
a234	Statutory sick pay - revng at present	set43	Variables used in derivation changed	
a239	Maternity allowance - no of wks rovd	set51	Variable used is now "bens2_numwks	
a240	Maternity allowance - rcvg at present	set51	Variable used is now "bens2_rcptnow	
a244	Hours paid overtime usually worked	set43	New raw variable "hourot" used	
a250	Mainjob - last wage was lst wk \month	set43	Variable "paymain_anticipay" not use	
a2551	1st subs job - empl prd cvrd by lastpay	set45	Variable "pay2_anticipay" not used	
a2552	2nd subs job- empl prd cvrd by lastpay	set45	Variable "pay2_anticipay" not used	
a256	FIS - no of weeks received	set51	Condition (cards1) changed (now = 3	
a257	FIS - receiving at present	set51	Condition (cards1) changed (now = 3	
a280	Maintenance / separation allwnce rcvd	set46	Formula changed to avrge of mntwho	
b010p	Rent\\rates - lastnet payment	set12	Variables/cndtns & formulae changed	

Family Expenditure Survey 1995-96 Derived variables amended in the 1996-97 dataset

Variable	Description of Variable	Table Name	Description of Change
b010u	Rent\\rates - last net payment	setunanonh	Variables/cndtns & formulae changed
b020p	Netrent - service charge deducted	set12	Variables/cndtns & formulae changed
b020u	Netrent - service charge deducted	setunanonh	Variables/condtns & formulae changed
b038u	Countax - last payment weekly amount	setunanonh	Raw variables used "dvawat & dvct1"
b050	Water charges - last net payment	set12	Variables/condition & formula changes
b053u	Council water tax - wkly amount paid	setunanonh	New variable ctwtpcnt inc in calculating
b150	Mortgage intrst \ pmcpl - interest paid	set12	Formula in condition 3 changed
b159	Service charge included in rent - amnt	set 15	Variables, formulae and notes changed
b165	Telephone bdgt a\c pmnt (ex mobiles)	set15	Calcited for landline/cable phones only
b166	Telephone, hhld shr of acc (ex mobiles)	set I 5	Calcited for landline/cable phones only
b167	Telephone bdget a\c chrge (ex mobiles)	set15	Calcited for landline/cable phones only
b172	Vehicle sold - net amount received	set l 4	Raw variable used "vsold howmuch"
ь175	Elec - amount paid in last account	set13	Formula/notes changed for new rebate
b1751	Eiec - amt pd in 1st accnt (all methods	set13	Coding frame deleted
b178	Elec - amount last slot meter rebate	set13	Rebate period changed to quarter!
b1781	Elec - amt of lst rebate (AMJ Qtr only)	set13	Rebate period changed to quarter!
Ь200	Mortgage intrst \\ prncpl - lst pymnt	set12	Mortgage type expanded in conditions
b222	Elec amnt of pymnt (brd bdgt scheme)	set13	Rebate period changed
b2241	Elec charge on lst bill (AMJ Qtr only)	set13	Change to quarter covered by variable
b225	Elec - windfall income Nat Grid rebate	set13	Local authority code variable changed
b230	Rent rebate - amt rcd in non-sep rte reb	set12	Variables used and formula changed
b244	Vehicle - cost of new car/van outright	set 14	Period covered and payment method
b246	Vehicle rd tx on cars curr owned last yr	set14	Period covered changed to 12 months
b247	Cost of motorcycles etc bt outright	set14	Vehicle type and payment method
b248	Car leasing - expenditure on	set14	Three new variables and formula
b252	Motorcycle - servs, repairs amt paid	set 14	Spare parts no longer included
b3071	Main job S/empld amount of net loss	set44	Variables used and formulae changed
b3072	1st subside job se amount of net loss	set44	Variables used and formulae changed
b3072	2nd subsid job se amount of net loss	set44	Variables used and formulae changed
b312	Bonus earnings - gross amount	set43	Tax rate used in calculation changed
b3262	First subs job se amount of net profit	set43	Major overhaul of derivation of b3262
b3263	Second subs job se amnt of net profit	set43	Major overhaul of derivation of b3263
b337	Child benefit last amount received	set51	Additional variable "dvonep" inc this yr
b342		set46	, -
b342	Statutory maternity pay - amount revd	set46	Matstrat deleted, 2 new variables Penstype changednowinclds 3,4,5&6
	Annuity\trust\covenant - last net pymnt		
b347	Annuity\trust\covenant - amnt tax ddctd	set46	Penstype changed now inclds 3,4,5&6
b348	Pension (public/private) last net pymnt	set47	Penstype changed now includes 1&2
b349	Pension (public\private) amnt tax ddctd	set47	Penstype changed now includes 1&2
b352	Rglr allwace frm o/s hh spouse/nonsp	set46	Table changed for allowance
b388	Statutory sick pay - imputed gross amnt	set49	Rate of sick pay increased to £61 00
b391	Self-employed tax paid last 12 months	set44	Variables / conditions and formulae
ctpent	Council Tax amnt pd as % of amt due	setctp	Only variables b029 & b038 now used
ctwtpcnt	Cncl wat chge amt pd as % of amt due	setctp	Council tax & water charge excluded

1995-96 Derived variables amended in the 1996-97 dataset Family Expenditure Survey Table Name Description of Change Description of Variable Variable Beer and lager - large supermarkets set27a Title changed from "large shops" d0401011 set27a Title changed from "large shops" d0401021 Cider - large supermarkets Title changed from "large shops" set27a d0401031 Table wines/wines nos - lrge sprmrkts set27a Title changed from "large shops" d0401041 Champagne/spkling wines - lg spimrkts set27a Title changed from "large shops" Fortified wines - large supermarkets d0401051 Title changed from "large shops" set27a d0401061 Spirits, Liquers - large supermarkets set27a Title changed from "large shops" d0401071 Alc sft drinks/alcopops - lg supermrkets set27a Title changed from "large shops" Off-licence Alcohol nos - lg spermrkets d0401081 set27a ~ Title changed from "large shops" d0702091 Electric consummables - Irge spermrkts Title changed from "large shops" set27a d0704011 Detergents etc - large supermarkets Title changed from "large shops" set27a Disinfectants etc - large supermarkets d0704021 Title changed from "large shops" set27a d0705011 Toilet paper - large supermarkets Title changed from "large shops" - set27a Petfood - large supermarkets d0706011 set27a Title changed from "large shops" Pet purchase etc -largesupermarkets d0706021 set27a Title changed from "large shops" Petrol - large supermarkets d1003011 Title changed from "large shops" set27a Diesel oil - large supermarkets d1003021 Title changed from "large shops" set27a d1003031 Other motor oils - large supermarkets New variable in place of SSRs set 1 Government Office Region gor First condition and 3 variables deleted set | 5c d080203c Telephone Account Condition/formula and variables used set900 d080206 Mobile phone account payments set900 Condition/formula and variables used Satellite TV subs to channels (diary) d130204 set43 Variable "dvins" Private medical -Gross wage, salary last time pd (main) p003 - insurance included in calculations Normal gross wage, salary set43 p007 Variable"ftpt" replaces "sehours" set41 p020 Person working set43 New variables for S/E - major changes Income from self - employment (main) p047 Tax rate changed from 25% to 20% set46 p048Income from investments Data for variable "b385" excluded p050 Income from other sources set46 set44 Tax rate change dvtaxd & dvsetax Income tax, pymnts less rfnds (13wkrl) p067 set46 Tax rate changed from 25% to 20% Tax dctd frm pnsions, invests, annuities p068 This is a new variable for 1996/97 SE NI contribns paid last 12 months set44 p076 New variables for S/E - specn chigd p079 Income tax payments less refunds set44 Tax rate changed from 25% to 20% set46 p080Building society acnts - interest & tax Tax rate changed from 25% to 20% Int & tax on bk, nt svngs, TESSA accs set46 p081 Tax rate changed from 25% to 20% p082 Interest & tax on gilt-edge & war loans set46 Tax rate changed from 25% to 20% Interest & tax on stocks and shares etc set46 p083 New benefit variables - specn changed Any other benefit - amnt currently rovd set51 p085 Expenditure code 031111 deleted p104 Food - creditexpenditure set54 New codes 040108 & 040208 set54 p105 Alcoholic drink - credit expenditure Expenditure code 031111 deleted set54 pll8 Food - personal expenditure Expenditure code 031111 deleted set54c pll8c Food - personal expenditure New codes 040108 & 040208 set54 Alcoholic drink - personal expenditure p119 New codes 040108 & 040208 Alcoholic drink - personal expenditure set54c p119c Expenditure code 080206 removed set54c Hhold services (RPI) - personal exptre p131c

Family Expenditure Survey

1996-97: SECTION 5, PART 6
1995-96 Derived variables amended in the 1996-97 dataset

Varıable ———	Description of Variable	Table Name	Description of Change
pl3lp	Hhold services (RPI) - personal exptre	set54	Expenditure code 080206 removed
p13Iu	Hhold services (RPI) - personal exptre	setunanonp	Expenditure code 080206 removed
p134	Fares etc (RPI) - personal expenditure	set54	New codes 110103 & 110104
p134c	Fares etc (RPI) - personal expenditure	set54c	New codes 110103 & 110104
p136	Leisure services (RPI) - personal exptre	set54	New codes 130206 & 140307
p136c	Leisure services (RPI) - personal exptre	set54c	New codes 130206 & 140307
р136грі	Leisure services (RPI)	set54	New codes 130206 & 140307
р136гріс	Leisure services (RPI)	set54c	New codes 130206 & 140307
p146p	Hhld services - credit expenditure	set54	Expenditure code 080206 removed
p146u	Hhld services - credit expenditure	setunanonp	Expenditure code 080206 removed
p149	Fares & oth travel costs - credit expend	set54	New codes 110103 & 110104
p152	Credit component of total pers expend	set54	New expenditure code 130206
p152u	Credit component of total pers expend	setunanonp	Adds in B166 and B1661
p250	Elec-slot meter payments less rebates	set13	Amended for National Grid rebates
p523p	Household services	set15	Added B1661 - Mobile phone account
p523rpip	Household services (RPI)	set15	Added B1661 - Mobile phone account
p523rpiu	Household services (RPI)	setunanonh	Added B1661 - Mobile phone account
p523u	Household services	setunanonh	Added B1661 - Mobile phone account
p525	Motoring expenditure (neg possible)	set l 4	B251 deleted & d100202 added
p537	ONS fuel, light and power	set24	Added b1781 to formula
p545	ONS motoring expenditure (ngs poss)	set24	B251 deleted & d100202 added
p555p	Household services - nat accs (anon)	set24	Added B1661 - Mobile phone account
p555u	Household services - nat accs (unanon)	setunanonh	Added B1661 - Mobile phone account
p557	Motoring - NAT ACCS	set24	B251 deleted & d100202 added
p558	Leisure services - NAT ACCS	set24	D130204 dltd, d130608/09/10 added
p558c	Leisure services - nat accs	set24c	D130206c, 130608c/609c/610c added
p559	Miscellaneous services - NAT ACCS	set24	D140307 added
p559c	Miscellaneous services - nat accs	set24c	D140307c added
regmet	Standard region with metropolitan areas	set l	Uses variable hhold_lacode (was lac)
t102	Spares and accessories	set26	B251 deleted & d100202 added
tll4	Bicycles, boats - purchases and repairs	set26	D110104 bicycle sprs & accsries added
t1310	Cash gifts, donations	set26	Variable d140307 added to formula
t133	TV, video, satellite rental, TVlicences	set26	D130204 deleted from formula
t139	Gambling payments	set26	Variables d130508, 509, 510 added
t1393	Lotteries (including National Lottery)	set26	Variables d130508, 509, 510 added
t13931	National lottery stakes & scratchcards	set26	Variables d130508, 509, 510 added
t15	Other payments recorded	set26	Variable t158 deleted from formula
t151	Life assurance, contrib to pension funds	set26	Variable d140106 added to formula
t1512	Contribs to pnsn / suprntn ded frm pay	set26	Variable d140106 added to formula
t311			
UII	Meat pies, sausages and other meats	set26	Variable d030512 added to formula
t340	•	set26 set26	Variable d030512 added to formula Variable d031111 added to formula
	Meat pies, sausages and other meats		

1996-97: SECTION 5, PART 6

1995-96 Derived variables amended in the 1996-97 dataset Family Expenditure Survey

1995-96	Derived variables amended in the 19	ramily Expenditure Survey		
Variable	Description of Variable	Table Name	Description of Change	
t83	Telephone	set26	D080206 deleted, b1661 added	
wkgross	Grosspay - amount	setwrk	Dvins (Priv Med Ins) added to formula	
wkly230	Carpamt - wkly equivalent value field	setwev	Period changed from 3 to 12 months	
wkły235	Cpparamt weekly equivalent value field	setwev	Period changed from 3 to 12months	
wklyvpin	Wey vehicle ins amt in purchase price	setwev	Period changed from 3 to 12 months	
wklyvptx	Wev roadtax amount in purchase price	setwev	Period changed from 3 to 12 months	

Expenditure Code Changes for 1996-97

1996-97: SECTION 5, PART 7 New Expenditure codes for 1996-97

Expenditure Code	Description of Expenditure Code
030512	Meat uncooked unspecified
040108	Alcohol from off-licence nos
040208	Alcohol consumed on licensed premises nos
110104	Accessories and repairs of bicycles etc
130206	Cable TV subscription
140106	Contributions to widows and orphans funds (deductions from pay)
140307	Residential care fees for the sick and elderly
100408	Motoring Fines and penalties
130508	National Lottery - saturday draw stakes
130509	National Lottery - mid-week draw stakes
130510	National Lottery - both weekly draw stakes
130608	National Lottery - saturday draw winnings
130609	National Lottery - mid-week draw winnings
130610	National Lottery - both weekly draw winnings

Family Expenditure Survey 1995-96 Expenditure codes that have been deleted for 1996-97

Expenditure Code	Description of Expenditure Code
<u> </u>	
031111	Foods other and undefined (now included in 031110)

Family Expenditure Survey 1995-96 Expenditure codes amended in the 1996-97 dataset

Expenditure Code	Description of Expenditure Code	Description of Change	
031110	Food undefined	Combines 031110 and 031111	
040107	Alcoholic 'soft drinks', alcopops from off-licence	Was alcohol from off-licence nos	
040207	Alcoholic 'soft drinks', alcopops consumed on licensed premises	Was alcohol consumed on licensed premises no	
060302	Women's accessories	Now includes headband, hairband - formerly in 060501 (Haberdashery)	
100205	Motor cycle accessories and spare parts	Spare parts moved from 100204 (Motor cycle repairs, service)	
110102	Boat house hire, outboard motor purchase	Now excludes bicycle purchase (which is now coded 110104)	
120114	Computer software	Now includes computer discs, computer head cleaners, formerly in 120108 (Records, CDs etc.)	
130205	Cable TV connection	Previously included subscription also (now 130206)	

Other Changes for 1996-97

Table Name	Description
ADETHNIC	Ethnic origin of each adult in the household
ALLOW	Receipt of money from other family member either in/out of household
BENRIAMT	Amount of benefit received for various DSS benefits
CHETHNIC	Ethnic origin of each child in the household
CTPAYS	This table contains details of council tax sewerage charges based on council tax band A. These charges are relevant to Scotland only. From 1997-8 this variable will be included in the hhold! raw table
HBSERVAM	Services in rent
HHOLDI	This household-level table contains data for most of the questions asked of all households, including the filter questions for the individual specific questions. There is an entry for each household. NOTE This table contains the first 106 variables of the hhold table which was split since it
	contained too many variables for SIR
HHOLD2	This household-level table contains data for most of the questions asked of all households, including the filter questions for the individual specific questions. There is an entry for each household. NOTE This table contains the second half of the hhold table which was was split since it contained too many variables for SIR.
PHONE	Telephone payments

Family Expenditure Survey

1995-96 Tables that have been deleted for 1996-97

Table Name	Description
BENTIAMT	This table has benefit-level details for those people receiving benefits on card T1 (eg funeral expenses grant) There is an entry for each type of benefit received which is indexed by 'CARDT1' which has the same coding frame as variable 'BNCARDT' in table 'BENT1' This enables a link to be made with the table 'BENTT1' which has the WEVs
EXPG	This transaction-level table contains details of money received from from outside the household to pay for any recorded expenditure. There is an entry for each item for which money is received. The filter is 'HHOLD_FILTER20' in table 'HHOLD'. 1=Yes, moneny rec'd.
ННЕХР	This item-level table contains details of household expenses paid direct by a spouse outside the household. There is an entry for each expense paid directly [The filter is 'SPSALDIR' in table 'VARIOUSI' 1=Yes, expenses paid]
HHOLD	This household-level table contains data for most of the questions asked of all households, including the filter questions for the individual specific questions. There is an entry for each household
PARTS	This vehicle-level table contains details of spare-parts bought for vehicles. There is an entry for each vehicle. The filter is SPARPT in table HHOLD. 1=Yes, spare parts bought.
RENTS2	This household-level table contains details about water/sewerage charges for rented accommodation. There is an entry for each renting household. The table RENTS1 contains the main renting information.
ROOMS	This room-level table contains details of the rooms in each household. There are 6 entries for each household, ie one entry for each type of room. The index variables are CASENO & ROOMS_TYPE

MAJOR CHANGES

1995-96 to 1996-97 Changes

Self-employment - The way in which information about income from self-employment is collected was substantially revised in 1996-97 following various tests and pilot studies. The quality of such data will be increased but this may lead to a discontinuity. See Section 5, Part 13 - Questionnaire Changes for full details

Cable/satellite television - Information on cable and satellite subscriptions is now collected from the household questionnaire rather than from the diary, leading to more respondents reporting this expenditure

Mobile phones - Expenditure on mobile phones was previously collected through the diary From 1996/97 this has been included in the questionnaire

Job seekers allowance (JSA) - Introduced in October 1996 as a replacement for Unemployment Benefit and any Income Support associated with the payment of Unemployment Benefit Receipt of JSA is collected with NI Unemployment Benefit and with Income Support In both cases the number of weeks a respondent has been in receipt of these benefits is taken as the number of weeks receiving JSA in the last 12 months and before that period the number of weeks receiving Unemployment Benefit/Income Support

Retrospective recall - The period over which information is requested has been extended from 3 to 12 months for vehicle purchase and sale. Information on the purchase of car and motorcycle spare parts is no longer collected by retrospective recall, instead expenditure on these items is collected through the diary

State benefits - The lists of benefits specifically asked about were reviewed in 1996/97 See the Income Questionnaire for more information

Sample stratifiers - New stratifiers were used in 1996/97 based on standard regions, socio-economic group and car ownership

Government Office Regions - Regional analyses are now presented using the Government Office Regions (GORs) formed in 1994 Previously all regional analyses used Standard Statistical Regions (SSRs)

Full questionnaire changes - please refer to Section 5, Part 13 for full details

Checking for changes

The checking of variables can be done using the different parts of this Section of the documentation Particular points to check are

Has the specification of the variable changed?

The formula from which a variable is calculated may have changed or the conditions on which that formula is based may have changed Specification changes can be confirmed using other parts of this section of the Documentation, particularly Parts 3 and 6 The structure of the 1996-97 product codes is given in more detail in Section 3, Part 4 of the Documentation Details of the specification of individual variables are available on request

Has the coding frame changed?

Changes to coding frames can significantly alter the distribution of data within a variable or a group of variables. It is very important that coding frame changes are noted. Changes to coding frames for raw variables can be confirmed using Parts 3 and 6, or Part 13 which shows coding changes in the context of the questionnaires.

Has the question wording changed?

This can be confirmed using Part 13

Questionnaire Changes

QUESTIONNAIRE CHANGES FROM APRIL 1996

Based on a paper by Social Survey Division

This paper lists the changes in the content of the FES questionnaires from April 1996. It covers new questions, deletions, amendments to wording, coding changes within specific questions on the Household and Income questionnaires and major routing changes.

The third column gives the 1995-96 question text and explains the nature of the change in the case of completely new questions a routing statement is included, e.g. APPLIES IF BUYING WITH MORTGAGE, followed by the question text Changes to coding or wording are indicated by "CODING CHANGE" or "REWORDED" followed by the new codes or wording

Questions are covered by topic area, e.g. mortgages

The full text and routing for all questions and the diary coding frame are available in separate documentation, see Section 6 - Variable Relationships

Changes to the questionnaire in 1996/97

The main changes to the FES questionnaire for 1996/97 are summarised below Most of the changes were made as a result either of the desire to harmonise questions asked on Government surveys or to make questions on the FES and Family Resources Survey (FRS) more similar A further major change was the introduction of a new section on self-employed income

MAIN QUESTIONNAIRE CHANGES

Standard Period Codes

A new period coding frame was introduced from April 1996. A number of questions relating to period covered use this standard set of codes

One week	1
Two weeks	2
Three weeks	3
Four weeks	4
Calendar month	5
Two calendar months	7
Eight times a year	8
Nine times a year	9
Ten times a year	10
Three months	13
Six months	26
A Year	52
Less than one week	90
One off/lump sum	95
None of these	97

N B codes 95 and 97 are for interview use only and will be re-coded to one of the other codes in the dataset

Household Questionnaire

- Information on household composition (for harmonisation)
 - a New question on date of birth of those aged under 20
 - b Separate questions on de facto and de jure marital status
 - c Identification of householder(s), 1e in whose name the accommodation is owned or rented
 - d Separate code for cohabitee in the relationship grid
 - e Introduction of a question on ethnic group (for each person)
 - f Age completed continuous full-time education (if aged 18+)
- 2 Restructuring of questions on length of residence, tenure and housing type to conform with the Survey of English Housing
- 3 Questions on number of rooms in accommodation revised to align with FRS
- 4 Rewording of the rent and housing benefit questions to align with FRS
- 5 Introduction of right to buy question Was accommodation previously rented and from whom
- Questions on repayment of interest-only mortgages amended to be specific to type of mortgage, e.g. pension plans for pension mortgage
- Redesigned consumer durable questions to cover additionally black and white televisions, satellite receivers, mobile phones and home computers
- 8 Council Tax questions
 - a Code introduced to allow CT questions to be off the route for household spaces not separately valued for CT
 - b Questions on CT transitional relief deleted

Questionnaire Changes

- 9 Telephone payment questions to identify separately landline, mobile phone and cable phone payments
- 10 Changes to questions on vehicles
 - a Reference period for questions on purchase and sale of vehicles extended to 12 months
 - b New question on car leasing payments on company cars
 - Questions on vehicle servicing and other labour costs to cover all vehicles owned or in continuous use in the previous 3 months
- New coding frame for payments covering different periods of time weekly, four weeks, calendar month, three months etc

Income Questionnaire

- I Government schemes for employment training expanded to include Work Trial (GB) and Job Skills (NI)
- New (opinion) question on whether those in employment work full or part time
- 3 New questions on hours of paid and unpaid overtime
- 4 New set of questions on income of the self-employed, which divide broadly into four sections which cover
 - a profit and loss for those who have prepared business accounts for tax purposes,
 - b drawings made from the business for those with a separate business/ work account,
 - c drawings made by those who do not have a separate work accounts (or do not draw from it),
 - d income tax payments and National Insurance contributions
- 5 Changes to benefits questions
 - a Child benefit separated from one-parent benefit
 - b Coding frame extended to cover Family Credit Lump Sum, Social Fund Grants for funeral and maternity expenses, and Community Care Grant separately (12 month reference period)
- 6 Information on pension income collected separately for different types of income from previous employer, private personal pension etc
- Information for regular allowances collected separately by type of allowance from absent household member, maintenance/ separation allowance, allowance for foster child etc

	Questionnaire	Changes
--	---------------	---------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		HOUSEHOLD QUESTIONNAIRE	
		Household composition (adults)	
Н_20 30	-	AgeDOB NEW QUESTION APPLIES IF Age < 20 As you are under 20, may I just check what is your/person's date of birth	า
H_20 35	H_34 6	MS REWORDED AND CODING CHANGE Are you/is (person)	
		Single, that is never been married Married and living with husband/wife Married and separated from husband/wife Divorced or Widowed	(1) (2) (3) (4) (5)
H_20 40	-	Living NEW QUESTION APPLIES IF Adult > 1 AND MS = 1,3,4,5 May I just check, are you living with someone in this household as a cou Yes No	(1) (2)
H_20 50	H_34 5	Adult_R CODING CHANGE Code relationship to (HOH) Spouse Cohabitee Son/daughter (incl adopted) Step-son/daughter Foster child Son-in-law/daughter-in-law Parent Step-parent Foster parent Parent-in-law Brother/sister (incl adopted) Step-brother/sister Foster brother/sister Brother/sister-in-law Grand-child Grand-parent Other relative Other non-relative	(1) (2) (3) (4) (5) (6) (7) (8) (9) (10) (11) (12) (13) (14) (15) (16) (17) (18)
H_20 70	-	FtEdChk NEW QUESTION APPLIES IF FtEd = 1 AND Age > 18 May I just check, have you been in continuous full-time education since ch Yes No	

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
Н_20 75	H_34 9	TEA REWORDED At what age did you/did complete continuous full-time education that period of education that started in early childhood?	at is, the
H_20 85		ADETHNIC_EthGrp NEW QUESTION APPLIES TO ALL AGED 16-99 To which of these groups do you consider you/person belongs? White Black-Caribbean Black-African Black - neither Caribbean or African Indian Pakistani Bangladeshi Chinese None of these	(1) (2) (3) (4) (5) (6) (7) (8) (9)
		Household composition (children)	
H_30 30	•	cAgeDOB NEW QUESTION APPLIES TO ALL AGED 0-15 What was child's date of birth?	
H_30 75	-	CHETHNIC_EthGrp NEW QUESTION APPLIES TO ALL AGED 0-15 To which of these groups do you consider child belongs? White Black-Caribbean Black-African Black - neither Caribbean or African Indian Pakistani Bangladeshi Chinese None of these	(1) (2) (3) (4) (5) (6) (7) (8) (9)
		Accommodation	
H_50 5	H_80	AcomType CODING CHANGE INTERVIEWER CODE IS THE HOUSEHOLD'S ACCOMMODATION a house or bungalow a room/rooms a flat or maisonette other	(1) (2) (3) (4)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_50 10	-	HseType NEW QUESTION APPLIES IF AcomType = I INTERVIEWER CODE IS IT detached semi-detached	(1) (2)
		or terraced/end of terraced?	(3)
H_50 15	-	FlatType NEW QUESTION APPLIES IF AcomType = 3 INTERVIEWER CODE IS IT IN a purpose built block a converted house/some other type of building?	(1) (2)
H_50 20	-	OthAcTyp NEW QUESTION APPLIES IF AcomType = 4 INTERVIEWER CODE IS IT a caravan, mobile home or houseboat some other kind of accommodation?	(1) (2)
H_50 25	-	BusDwell NEW QUESTION APPLIES TO ALL INTERVIEWER ASK OR CODE WHETHER ADDRESS IS ATTACHED BUSINESS PREMISES	
		Yes No	(1) (2)
H_50 30	H_82	LtResYrs REWORDED AND CODING CHANGE For how many years have you lived at this address? Less than 12 months 12 months but less that 2 years 2 years but less than 3 years 3 years but less than 5 years 5 years but less than 10 years 10 years or longer	(1) (2) (3) (4) (5) (6)
н_50 50	-	Sublet NEW QUESTION APPLIES TO ALL Do you have a formal arrangement to let or sublet any part of this accommod to someone who is NOT a member of your household? Yes No	dation (1) (2)
H_50 55	-	SubLety NEW QUESTION APPLIES IF Sublet = 1 Who is that? Close relative Other relative Non-relative	(1) (2) (3)
Н_50 60	-	Rooms NEW QUESTION APPLIES TO ALL How many rooms do you have in your accommodation, that is excluding bath and toilets but including kitchens?	irooms

1996-97 question	1995-96 question	
number	number	1996-97 Questionnaire text or description of change
H_50 65	-	RoomShar NEW QUESTION APPLIES TO ALL Are any of these rooms shared with anyone who is not a member of your household
H_50 70	-	Bedroom NEW QUESTION APPLIES TO ALL (Thinking just of the accommodation occupied by your household) How man bedrooms do you have in this accommodation?
H_50 75	-	BusRoom NEW QUESTION APPLIES TO ALL Are any of the rooms you have mentioned used for business because you are se employed? Yes No (1)
H_50 80	-	OnBsRoom NEW QUESTION APPLIES IF BusRoom = 1 How many rooms are used wholly for business?
H_50 85	-	PtBsRoom NEW QUESTION APPLIES IF BusRoom = 1 How many rooms are used partly for business?
		Type of tenure
H_60 5	-	TENURE_HHOLDE00 TENURE_HHOLDE10 NEW QUESTION APPLIES TO ALL In whose name is the property owned or rented? 1 10 (household members) 97 (non-household member) Multi-coded up to 11 maximum
H_60 10	H_283	OwnRent REWORDED AND CODING CHANGE In which of these ways do you occupy this accommodation? Owns it outright Buying it with the help of a mortgage or loan Pay part rent and part mortgage (shared ownership) Rent it Live here rent free (including rent free in relative's/ friend's property, excluding squatting (5) Squatting
H_60 15	H_286	INA110 REWORDED Does the accommodation go with the job of anyone in the household? Yes No (1) No

Questionnaire	Changes
---------------	---------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_60 20	H_284	Landlord REWORDED AND CODING CHANGE Who is your landlord? Organisations the local authority/council/New town development/ Scottish Homes/Northern Ireland Housing Executive	(1)
		a housing association or co-operative or charitable trust employer (organisation) of a household member another organisation	(2) (3) (4)
		Individuals relative/friend (before you lived here) of household member employer (individual) of a household member another individual private landlord	(1) (2) (3)
H_60 25	-	Furnish NEW QUESTION APPLIES IF Ownrent = 3, 4 or 5 Is the accommodation provided	
		Furnished Partly furnished (e.g. only carpets and curtains) Unfurnished	(1) (2) (3)
		Rent and housing benefit	
		1995/6 QUESTIONS 289 TO 322A DELETED	
H_70 35	•	HBenefit NEW QUESTION APPLIES IF OwnRent = 3 or 4 Some people qualify for Housing Benefit, that is, a rent rebate or allow you receiving Housing Benefit, either directly or by having it paid to you on your behalf?	
		Yes No	(1) (2)
H_70 40	-	Rebate NEW QUESTION APPLIES IF Rent = 0 AND HBenefit = 1 You said that you paid no rent last time Is that because you get 100% Benefit?	6 Housing
		Yes No	(1) (2)
H_70 45	-	Rebate0 NEW QUESTION APPLIES IF Rent = 0 AND HBenefit = 2 OR Reba Can I just check, what is the reason for you paying no rent last time?	te = 2
H_70 50	-	HBenAmt NEW QUESTION APPLIES IF HBenefit = 1 How much rent rebate (Housing Benefit) were you allowed last time?	
H_70 55	-	PercHBen NEW QUESTION APPLIES IF HBenAmt = 0 01-9997 00 What period does this cover?	

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_70 65	Н_308	Rentbar REWORDED May I check, was your last rent payment, that is £(t/s amount at Rent), before after the Housing Benefit/rent rebate was deducted? Before After	(1) (2)
		Services in rent	
H_70 70	-	HBStmt NEW QUESTION APPLIES IF HBenefit = 1 For people on Housing Benefit, the Council sends you a statement to show how the amount of benefit was calculated Do you have a statement from Council, or any other document such as a rent card, which you could consult Yes	n the t? (1)
		No	(2)
H_70 75	-	HBWater NEW QUESTION APPLIES IF HBStmt = 1 AND LACode = 1 402 {England Wales} On the statement, is there an amount for water charges? Yes No	d and (1) (2)
H_70 80	-	HBSewer NEW QUESTION APPLIES IF HBStmt = I AND LACode = 1 402 And on the statement is there an amount for sewage charges? Yes No	(1) (2)
Н_70 85	-	HBWsAmt NEW QUESTION APPLIES IF HBWater = 1 OR HBSewer = 1 What is the amount for water/sewerage?	
H_70 95	-	HBSERV00 HBSERV11 NEW QUESTION APPLIES IF HBStmt = 1 And on your document, are there amounts shown for any of the things show card C1?	n on
		Heating Lighting Hot Water Fuel for cooking Food, including prepared meals TV/Video rental TV Licence fees Personal laundry Medical expenses Nursing and personal care Transport Contents insurance None of these	(1) (2) (3) (4) (5) (6) (7) (8) (9) (10) (11) (12) (13)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_70 105	-	HBSerAmt NEW QUESTION APPLIES IF HBServ = 1-12 How much is the amount on the statement for (each service)?	
Н_70 115	-	HBSvOth NEW QUESTION APPLIES IF HBStmt = 1 Are there any other services in the rent we have not covered? Yes No (1) (2)	
Н_70 120	-	HBSOther NEW QUESTION APPLIES IF HBSvOth = 1 What is covered by these deductions?	
H_70 125	-	HBSOAmt NEW QUESTION APPLIES IF HBSOth = Response What is the total amount deducted for this/these service(s)?	
Н_70 135	-	Worsinc NEW QUESTION APPLIES IF (LACode = 1 to 402) AND (Rent > 0 or DK refusal) AND ((HBWater = 1 AND HBSewer <> 1) OR (HBWater <> 1) AN (HBSewer = 1)) Were water/sewerage charges included in the rent which you mentioned? Yes No (1)	i)
H_70 140	-	WSInc NEW QUESTION APPLIES IF (LACode = 1 to 402) AND ((Rent > 0 or DK Refusal) AND HBStmt <> 1) OR (HBWater <> 1 and HBSewer <> 1)) Were water or sewage charges included in the rent which you mentioned? Both water and sewerage Water only Sewerage only Neither	1) 2) 3)
H_70 145	-	WSIncAmt NEW QUESTION APPLIES IF WSInc = 1, 2, or 3 OR WorsInc = 1 How much was included?	
H_70 155	-		

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_70 160	-	SerInc00, SerInc01 SerInc12 NEW QUESTION APPLIES IF SerInc = 1	
		Which services?	4
		Heating Lighting	(1)
		Lighting Hot Water	(2) (3)
		Fuel for cooking	(4)
		Food, including prepared meals	(5)
		TV/Video rental	(6)
		TV Licence fees Personal laundry	(7) (8)
		Medical expenses	(9)
		Nursing and personal care	(10)
		Transport Contents insurance	(11) (12)
		Other services	(12)
			` ,
H_70 165	-	OthSrv	
		NEW QUESTION APPLIES IF SerIncW = 13 What is/are this/these other services?	
		Trial is are mischiese only services	
H_70 170	-	ServAmt	
		NEW QUESTION APPLIES IF SerInc = 1 How much in total was included for the services in the rent which you mention	onad?
		How much in total was included for the services in the fent which you mention	Jileu ,
		Water charges not paid as part of rent	
		1995/6 QUESTIONS 340 TO 354 DELETED	
H_90 5	-	WaterPay	
		NEW QUESTION APPLIES IF LACode = 1 402 {England and Wales} AND (((WSInc = EMPTY) AND (WorSinc = EMPTY) AND (HBWater <> AND (HBSewer <> Yes)) OR (WSInc = 3 or 4) OR ((WorSinc = No)	
		(HBWater <> Yes)))	
		Do you pay water rates? Yes	(1)
		No, not connected or paid direct by someone else	(2)
		Bill not yet at this new address	(3)
H_90 10	-	SewerPay NEW QUESTION APPLIES IF LACode = 1 402 {England and Wales} and (((WSInc = EMPTY) AND (Worsinc = EMPTY) AND (HBWater < > AND (HBSewer < > Yes)) OR (WSInc = 2 or 4) OR (WorSinc = No (HBSewer < > Yes)))	
		Do you pay sewerage rates or charges?	
		INCLUDE CHARGE FOR EMPTYING SEPTIC TANK	/ 115
		Yes No charges hard or hard direct by someone else	(1) (2)
		No charges paid or paid direct by someone else Bill not yet paid at this address	(3)
			` '

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change
H_90 15	-	SewSep NEW QUESTION APPLIES IF WaterPay = 1 AND SewerPay = 1 Do you pay separate or combined water and sewerage rates? Separate Combined (1)
		Water charges
H_90 20	-	WatAmt NEW QUESTION APPLIES IF SewSep = 1 OR (WaterPay = 1 AND SewerPay = 2) How much did you actually pay last time for water rates?
H_90 25	-	PercWat NEW QUESTION APPLIES IF SewSep = 1 OR (WaterPay = 1 AND SewerPay = 2) What period did this cover?
H_90 40	-	SewAmt NEW QUESTION APPLIES IF SewSep = 1 OR (WaterPay = 2 AND SewerPay = 1) How much did you actually pay last time for sewerage rates?
H_90 45	-	PercSew NEW QUESTION APPLIES IF SewSep = 1 OR (WaterPay = 2 AND SewerPay = 1) What period did this cover?
H_90 60		WSewAmt NEW QUESTION APPLIES IF SewSep = 2 How much did you pay last time?
H_90 65	-	PercWSew NEW QUESTION APPLIES IF SewSep = 2 What period did this cover?
		<u>Mortgages</u>
H_110 30	-	Exrent NEW QUESTION APPLIES IF OwnRent = 2,3 OR BuyYear > 80 May I check, had you been renting this house/flat before deciding to buy it? Yes No (1)
H_110 35	-	Rentfrom NEW QUESTION APPLIES IF Exrent = I Who was it rented from? Local Authority or Council Housing Association/co-op/charity Employer Other organisation Other individual (5)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_110 45	H_387	bank	(1) (2) (3)
H_110 60	Н_391	MortLen REWORDED In which year did your current mortgage begin?	
H_110 70	Н_393	MortPay REWORDED How much was your last payment on this mortgage or loan?	
H_110 95	н_396В	Person 2 Person 3 Person 4 Person 5 Person 6 Person 7 Person 8 Person 9	(1) (2) (3) (4) (5) (6) (7) (8) (9) (10)
H_110 110	Н_397		the (1) (2)
		Mortgage endowment policies	
H_110 120	Н_398В	MenPolam REWORDED Policy (number) /How much was your last premium on the endown policy/PEP/unit trust/pension plan?	nent
H_110 140	H_402	Menstyr REWORDED Policy (number) /In what year was this insurance/PEP/policy taken out?	
H_110 145	H_403		ering (1) (2)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Mortgages	
H_110 165	H_429	IntMnth REWORDED How many months does this cover?	
		1995/6 QUESTIONS 430 TO 432 DELETED	
		Mortgage protection policies	
H_120 40	Н_436	IncMP REWORDED Is this payment included in the amount of £ you mentioned earlier? Yes No	(1) (2)
		Council Tax	(2)
H_150 5	H_488A	CTCondoc REWORDED The next few questions are about Council Tax Do you have a statement of from the council that we could look at?	
		Yes No	(1) (2)
H_150 10	H_488B	CTband REWORDED AND CODING CHANGE Could you tell me in which Council Tax band this accommodation is in? Band A Band B Band C Band D Band E Band F Band G Band H Household accommodation not valued separately	(1) (2) (3) (4) (5) (6) (7) (8) (9)
H_150 15	H_489	CTdisab REWORDED Was your Council Tax bill reduced to a lower band because there is a di person in your household? Yes No	(1) (2)
H_150 25	Н_490	CTAmt REWORDED How much Council Tax did you actually pay last time (including for council charge), after deducting any discounts or benefits?	l water

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change
Н_150 35	H_491	CTWat NEW QUESTION APPLIES IF FESCODE = 403 - 455 {Scotland} AND CTAmt > 0 How much was included in your last Council Tax payment, that is , for council water charge?
H_150 55	-	CTIrreg NEW QUESTION APPLIES IF CTInstal = 3 Please explain when amounts are paid and how often?
H_150 60	H_490D	CTAnnual REWORDED On the statement/bill what is the total amount payable for the whole year (including for council water charges) after deducting any discounts or benefits?
Н_150 65	-	CTWatAnn NEW QUESTION APPLIES IF FESCODE = 403 - 455 {Scotland} AND CTCondoc = 1 On the statement what is the total amount payable for the whole year for council WATER charge only?
		Private pension contributions
H_160 45	-	PensChk NEW QUESTION APPLIES IF Filter = 1 AND MortType = 3 Pension (number) /May I check, is this the same personal pension as the one covering your mortgage repayments? Yes No (1) (2)
		Life insurance
H_170 5	Н_527	HHOLD_Filter02 REWORDED In the last 12 months since (date), have (any of) you paid any premiums on life, death or endowment policies, or any annuity? Yes No (1)
H_170 50	н_535	LIFE_Anymore REWORDED In the 12 months since (date), have (any of) you paid any premiums on any other life, death or endowment policies, or annuities?
		Yes (1) No (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Medical/other insurances	
H_180 5	Н_592	HHOLD_Filter03 REWORDED In the 12 months since (date), have (any of) you made any payments on insurance policies such as those on card E1?	
		Yes No	(1) (2)
H_180 20	Н_594	Minstype CODING CHANGE Policy (number) /What type of policy is this?	(=)
		Personal accident policies	(1)
		Private medical insurance schemes (eg BUPA, HCS, PPA, PPP, WPA	(2)
		Permanent health insurance (PHI) Friendly societies sickness insurance (eg Hospital Savings	(3)
		Association	(4)
		Nursing home/Long-term care	(5)
		Cover for stay in hospital/other medical insurance Redundancy policy not covering mortgage	(6) (7)
		Credit card protection insurance	(8)
		Animal insurance	(9)
		Other	(10)
H_180 50	H_600	MEDINS_Anymore REWORDED Have (any of) you paid any more payments on policies such as those card E1?	shown on
		Yes	(1)
		No	(2)
		Consumer durables	
		NEW QUESTIONS REPLACE H_689	
H_190 5	-	Does your household have any of the following in your part of the accom-	modation?
		TVCol NEW QUESTION APPLIES TO ALL SPENDERS Colour TV	
		One	(1)
		More than one None	(2) (3)
H_190 10	-	TVbw	
		Black and white TV	(1)
		One More than one	(1) (2)
		None	(3)

1996-97: SECTION 5, PART 13 Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_190 15	-	SatDish Satellite Dish Yes No	(1) (2)
H_190 20	-	Video Video recorder Yes No	(1) (2)
H_190 25	-	Freezer Deep freezer of fridge freezer Yes No	(1) (2)
H_190 30	-	Fridge Separate refrigerator Yes No	(1) (2)
H_190 35	-	WashMach Washing machine Yes No	(1) (2)
H_190 40	-	Drier Tumble drier Yes No	(1) (2)
H_190 45	-	DishWash Dish washer Yes No	(1) (2)
H_190 50	-	MicroWve Microwave oven Yes No	(1) (2)
H_190 55	-	Telephon Telephone Yes No	(1) (2)
H_190 60	-	Mobile Mobile phone Yes No	(1) (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_190 65	-	CDPlayer COMPACT DISC Player Yes No	(1) (2)
H_190 70	-	Computer Home Computer Yes No	(1) (2)
		Telephone payments	
H_200 10	-	TelFilt NEW QUESTION APPLIES IF Telephon = 1 OR Mobile = 1 The next questions are about telephone accounts paid by you (or other members this household) May I first check, does your household pay for at least telephone account or are all your phone bills paid direct by someone else? HHld pays at least one account Whole bill paid by someone else SPONTANEOUS- Does not pay for telephone use	(1) (2) (3)
H_200 15	-	Person NEW QUESTION APPLIES IF Telp = 1 OR TelFilt = 1 Who pays this account?	
H_200 20	-	Phone NEW QUESTION APPLIES IF Telp = 1 OR TelFilt = 1 Account number	
H_200 25	-	TelType NEW QUESTION APPLIES IF Telp = 1 OR TelFilt = 1 What type of phone service is this? BT/Mercury landline Mobile phone Cable phone	(1) (2) (3)
н_200 30	-	Telehow NEW QUESTION APPLIES IF Telp = 1 OR TelFilt = 1 By which method do you pay? Account Budget scheme or by some other method	(1) (2) (3)
H_200 125	-	Anymore NEW QUESTION APPLIES IF TeleHow = 1, 2, or 3 Do any of you pay for any more telephone accounts? Yes No	(1) (2)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Electricity payments	
210 15	-	ElecRebt NEW QUESTION APPLIES IF ElecPay = 1,2,3,4,5,6,7,9, (In the 3 months since (date)) (In connection with your last electricity statement), have you received or been allowed any rebate, voucher or crevyour electricity company either because of the flotation of the National Gresome other reason such as a previous overpayment? Yes, rebate received No rebate received	dıt from
210 75	-	EChgamt NEW QUESTION APPLIES IF ElecPay = 2,3,5,6,7 or 9 How much was the CHARGE on your last electricity bill or statement? The is the amount shown for units consumed and the standing charge an excluding any amount for rental appliances, hire purchase, loans or maintenance charges?	d VAT,
		TV, video, satellite rental and subscription	
H_240 5	H_695	HHOLD_Filter04 REWORDED AND CODING CHANGE Do any of you make any of the rental or subscription payments shown on in connection with a TV, Video recorder, satellite channels or cable serving Yes No	
H_240 20	н_697	TVType REWORDED AND CODING CHANGE What does your payment cover? Hire of a TV only Hire of a video recorder only Hire of a TV set and a video recorder Hire of a TV set and subscription to satellite channels Hire of a TV set, Video and subscription to satellite channels Subscription to satellite channels only Cable TV including telephone Cable TV not including telephone	(1) (2) (3) (4) (5) (6) (7) (8)
H_240 45	-	TV_Anymore NEW QUESTION APPLIES IF Filter = 1 Do you make any more rental or subscription payments for any more of the on card F1?	ne things
		Yes No	(1) (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Owned and leased vehicles	
H_260 5	H_726	Filter 1 REWORDED At any time in the last 12 months since (date), have (any of) you OWN PRIVATELY LEASED any motor vehicles? Please remember to incivehicles that you have sold? Yes No	NED OR lude any (1) (2)
H_260 25	H727 2	VOWNED_Vehtype CODING CHANGE Is/was it a car (includes three wheel vehicles and normal cars converted for invalid use van (includes light vans, pick ups, landrover and jeeps) motorcycle or a moped (included scooters, invalid tricycles) or other motor vehicle	(1) (2) (3) (4) (5)
Н_260 75	H_727 12	Own REWORDED Is the privately owned or privately leased at present? Privately owned at present Privately leased at present Does not own or lease at present	(1) (2) (3)
		Vehicles bought	
H_270 25	H _729 2	VBOUGHT_VehType CODING CHANGE Was it a car (includes three wheel vehicles and normal cars converted for normal invalid use van (includes light vans, pick ups, landrover and jeeps) motorcycle or a moped (includes scooters, invalid tricycles) or other motor vehicle	(1) (2) (3) (4) (5)
Н_270 35	H_729 4	PayMeth CODING CHANGE Did you pay by loan or part cash / part loan from finance house, bank, building society Hire Purchase loan from employer credit card or cash, overdraft or a loan from a friend or relative?	(1) (2) (3) (4) (5)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Vehicles sold	
H_280 5	н_730	Filter3 REWORDED In the 12 months since (date), have (any of) you sold any of the vehicles mentioned, excluding trade-in or part exchange? Yes No	s you have (1) (2)
H_280 25	н_731 2	VSOLD_VehType CODING CHANGE Was it a car (includes three wheel vehicles and normal cars converted for invalid use) van (includes light vans, pick ups, landrover and jeeps motorcycle or a moped (includes scooters, invalid tricycles) or other motor vehicle	(1) (2) (3) (4) (5)
		Company / continuous use vehicles	
H_290 25	H_733 2	VUSED_Vehtype CODING CHANGE Is / was it a	(1) (2) (3) (4) (5)
H_290 80	-	CLea NEW QUESTION APPLIES IF Occont = 1 Do you yourself make any leasing payments for the , including any deducted from your salary? Yes No	y amounts (1) (2)
Н_290 85	-	CLeaAmt NEW QUESTION APPLIES IF Clea = 1 How much was your last car leasing/hire payment?	
H_290 95	-	PercCLea NEW QUESTION APPLIES IF Clea = 1 What period did this cover?	

Questionnaire C	Changes
-----------------	---------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_290 135	Н_733 17В	Reglet CODING CHANGE Please give the registration letter P N M L K J H G Other registration number	(1) (2) (3) (4) (5) (6) (7) (8) (9)
		Vehicle servicing	
H_3105	н_760	FiltSr REWORDED In the last six months have any of you paid for a service for any of the you own? Yes No	vehicles (1) (2)
H_310 30	Н_766 2	VehSrv CODING CHANGE Is the vehicle a car (includes three wheel vehicles and normal cars converted for invalid use van (includes light vans, pick ups, landrover and jeeps motorcycle or a moped (includes scooters, invalid tricycles) or other motor vehicle	(1) (2) (3) (4) (5)
		Other work on vehicles	
H_320 25	H_764 2	OTHW_Vehtype CODING CHANGE Is it a car (includes three wheel vehicles and normal cars converted for invalid use van (includes light vans, pick ups, landrover and jeeps motorcycle or a moped (includes scooters, invalid tricycles) or other motor vehicle 1995/6 QUESTIONS 765 TO 762 5 (vehicle spare parts) DELETED	(1) (2) (3) (4) (5)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Central heating installation	
400 5	Н_2908	CHIFilt REWORDED In the 12 month since (date), have (any of) you paid a contracto install central heating? Please exclude the cost of any items you a shop Yes	
		No	(2)
H_400 35	Н_2914	CHI_How CODING CHANGE Did you pay by loan Hire Purchase credit card Cash or other method?	(1) (2) (3) (4)
		Central heating maintenance	
H_410 5	Н_2940	CHMFilt REWORDED In the 12 months since (date),have (any of) you paid a contract for repairs, servicing or maintenance of central heating? Please you bought direct from a shop Yes No	
H_410 30	Н_2946	CHM_How CODING CHANGE Did you pay by loan Hire Purchase credit card cash or other method?	(1) (2) (3) (4)
		Capital Improvements	
H_420 5	н_2972	IMPFilt REWORDED In the 12 months since (date), have (any of) you paid a contract for any of these improvements or extensions? Please exclude any direct from a shop Yes No	

Questionnaire	Changes
---------------	---------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_420 30	H_2978	IMP_How CODING CHANGE Did you pay by loan Hire Purchase credit card cash or other method?	(1) (2) (3) (4)
		House maintenance	
H_430 5	Н_3004	DECFilt REWORDED In the 12 months since (date), have (any of) you paid a contractor or anyon for any maintenance, decorations, repairs or replacements such as those sho card H2? Please exclude the cost of any items you bought direct from a since Yes No	wn on
H_430 30	Н_3010	DEC_How CODING CHANGE Did you pay by loan Hire Purchase credit card cash or other method?	(1) (2) (3) (4)
		Moving house fees	
H_440 5	Н_3060	FeeFilt REWORDED In the 12 months since (date), have (any of) you paid any conveyancing agents' or surveyors' fees? Yes No	(1) (2)
H_440 25	н_3039	HEXP2_Amount REWORDED Property fee (number) In the last 12 months since (date), how much in t you pay, excluding stamp duty, land registry fees, local authority search frany interest paid on loans?	otal did ees and
		<u>Furniture</u>	
H_460 20	H_3117	FurnHow CODING CHANGE Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		<u>Carpets</u>	
H_470 20	H_3122	RugHow CODING CHANGE Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)
		Package holidays	
H_480 55	-	PackHow NEW QUESTION APPLIES IF Hpayamt = 0 9997 Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)
H_480 60	-	PackLHP NEW QUESTION APPLIES IF PackHow = 1 or 2 Are you still paying instalments on the loan/hire purchase? Yes No	(1) (2)
H_480 130	•	InsHow NEW QUESTION APPLIES IF SepInsAm = 1 997 Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)
H_480 135	-	InsLHP NEW QUESTION APPLIES IF InsHow = 1 or 2 Are you still paying instalments on the loan/hire purchase? Yes No	(1) (2)
		Hotel holidays	
H_490 40	-	HOLHOTL How NEW QUESTION APPLIES IF Hpaymt = 0 9997 Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)

Questionnaire C	hanges
-----------------	--------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
H_490 45	-	HOLHOTL LHP NEW QUESTION APPLIES IF How = 1 or 2 Are you still paying instalments on the loan/hire purchase? Yes No	(1) (2)
		Self-catering holidays	
H_500 40	-	HOLSCAT_How NEW QUESTION APPLIES IF Hpaymt = 0 9997 Did you pay by loan Hire Purchase credit card or cash or other method	(1) (2) (3) (4)
H_500 45	-	HOLSCAT_LHP NEW QUESTION APPLIES IF How = 1 or 2 Are you still paying instalments on the loan/hire purchase? Yes No	(1) (2)
H_510 5	H_3130A	Flights from UK Flyfilt REWORDED In the 12 months since (date), have any of you paid for a flight from a for holiday or leisure purposes (including any flights covered at the question about holidays? Yes No	
Н_520 20	Н_3133	Bank charges BkServCh REWORDED Account (number) Have you paid any bank SERVICE CHARGES on in the 3 months since (date) excluding payments for overdrafts? Yes No	this account (1) (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change
		Money given to household members
Н_650 5	-	HHOLD_ExpTyp00 HHOLD_ExpTyp07 NEW QUESTION APPLIES TO ALL SPENDERS Does anyone from outside this household GIVE YOU THE MONEY to pay for any of the items of household expenditure you have mentioned? Housing expenditure (1)
		Fuel expenditure (2)
		Vehicle expenditure (3)
		Other travelling costs (4)
		TV rental / licence (5)
		Video rental (6)
		School fees (7)
		Any other items of household expenditure (8) None of these (9)
11 (50.10		
H_650 10	•	HHOLD_ItemTypO NEW QUESTION APPLIES IF Filter = 8 What was the other expenditure?
H_650 15	-	GivenBy NEW QUESTION APPLIES IF Filter = 1-8 Who gave you the money for this expenditure?
H_650 20	-	HHOLD_ExpAmt NEW QUESTION APPLIES IF Filter = 1-8 How much in total did give you for this/these items?
		now much in total did give you for this/these nems.
H_650 25	-	HHOLD_ExpPC NEW QUESTION APPLIES IF ExpAmt = 0.01 9997 00 What period did this cover?
		1995/6 QUESTIONS 3608 TO 3617 DELETED
		INCOME QUESTIONNAIRE
		Employment Status
I_10 5	1_10 5	GovtProg REWORDED
		In the 12 months since (date), have you taken part in any of the governmen schemes for employment training shown on card 01?
		Yes (1) No (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
1_10 10	I_10 10	ProgType CODING CHANGE Which programme was that?	
		Training for work (GB) Youth Training (YT) (GB) Community Action (GB) Work Trial Enterprise Ulster/Job Training Programme (NI) Youth Training Programme (NI) Action for Community Employment (NI) Job Skills (NI) Other government Programme	(1) (2) (3) (4) (5) (6) (7) (8) (9)
I_10 25	I_10 25	TrainBen REWORDED What was the amount of allowance, including any payments from your e that you last received?	employer,
I_10 40	[_10 40	INA273 REWORDED Are you (receiving this allowance) (on work trial) at present? Yes No	(1) (2)
i_10 45	I_10 45	Paidwork REWORDED Did you do any paid work in the 7 days ending last Sunday either as an eor as self-employed? Yes No	(1) (2)
I_10 46	-	Awayfrom NEW QUESTION APPLIES IF Paidwork = 2 Did you have a job or business you were away from? Yes No SPONTANEOUS Waiting to take up a job already obtained	(1) (2) (3)
I_10 50	1_10 50	OwnBiz REWORDED Did you have any unpaid work in a business that you own? Yes No	(1) (2)
1_10 60	1_10 60	Looked REWORDED Thinking of the four weeks ending last Sunday were you looking for an paid work or government training scheme at any time in those 4 weeks? Yes No	
	I_10 65	LookYT DELETED	(~)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_10 70	I_10 70	Waiting REWORDED Were you waiting to take up a job that you had already obtained? Yes No	(1) (2)
1_10 71	I_10 80	Starty REWORDED If a job or a place on a government scheme had been available in the we last Sunday would you have been able to start within 2 weeks? Yes No	
I_10 75	I_10 75	YNotLk REWORDED What was the main reason you did not seek any work in the last 4 weeks/ be able to start work in the next 2 weeks? Student Looking after the family/home Temporarily sick or injured Long-term sick or disabled Retired from paid work None of these	(1) (2) (3) (4) (5) (6)
	I_10 80 I_10 95	StartJ DELETED AtWork3 DELETED	
	I_10 100	Away3 DELETED	
1_10 105	I_10 105	INA207 CODING CHANGE What is the reason for your absence illness or accident holiday strike maternity leave paternity leave pattern of shifts laid off compassionate leave or some other reason	(1) (2) (3) (4) (5) (6) (7) (8) (9)
I_10 120	I_10 120	DtlstWkY REWORDED When did you leave your last paid job?	
1_10 125	1_10 125	DtLstWkM REWORDED Which month in that year did you leave?	

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
<u>number</u>	Hamber	1990-97 Questioniane text of description of change	
		Job Description	
I_20 I	-	JOBMAIN_INDTITLE NEW QUESTION APPLIES IF Paidwork = 1 OR DtistwkY > 20 OR Awayfrom = 1 Please enter a short title for industry	
[_20 15	I_20 I5	JOBMAIN_JOBTITLE REWORDED What was your (main) job?	
I_20 31	-	JOBMAIN_ETYPE NEW QUESTION APPLIES IF Paidwork = 1 OR DtlstwkY >20 OR AWAYFROM = 1 Please look at this card and tell me which of these best describes your employment situation in this job Employee Running a business or professional practice Partner in a business or professional practice Working for myself	(1) (2) (3) (4)
		A subcontractor (includes SC60) Doing freelance work Self-employed in some other way	(5) (6) (7)
1_20 36	-	JOBMAIN_DIRNI NEW QUESTION APPLIES IF Director = 1 In this job, are/were your National Insurance contributions deducted at so Yes No (including no contributions paid)	ource? (1) (2)
I_20 40	I_20 40	JOBMAIN_WORKSTAT No longer asked Now a derived variable Code 1 if EType = 1 Else code 2	
1_20 45	1_20 45	Supervis CODING CHANGE Did you have any managerial duties, or were you supervising any other employees? manager foreman/supervisor not manager/supervisor	(1) (2) (3)
I_20 50	1_20 50	SizeEst CODING CHANGE How many employees were there in the place where you worked? 1-24 25 or more	(1) (2)
1_20 65	-	WorkYr NEW QUESTION APPLIES IF PAIDWORK = 1 OR AWAYFROM = In which year did you start working continuously for your current emple self-employed person?	l oyer/as

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
1_20 70	-	FtPt NEW QUESTION APPLIES IF Paidwork = 1 OR Awayfrom = 1 In your job are you working Full time Part time	(1) (2)
I_30 6	-	JOB2_Indtitle NEW QUESTION APPLIES IF MtOneJob = 1 Please enter a short title for industry	
I_30 26	-	JOB2_ETYPE NEW QUESTION APPLIES IF MtOneJob = 1 Please look at this card and tell me which of these best describes your employment situation in this job? Employee Running a business or professional service Partner in a business or professional practice Working for myself A subcontractor (including SC60) Doing freelance work Self-employed in some other way	(1) (2) (3) (4) (5) (6) (7)
I_30 31	-	JOB2_DirNI NEW QUESTION APPLIES IF Director = 1 In this job, are/were N1 contributions deducted at source? Yes No	(1) (2)
-	1_30 35	JOB2_WORKSTAT No longer asked Now a derived variable Code 1 if EType = 1 Else code 2	
I_40 6	-	JOB3_Indtitle NEW QUESTION APPLIES IF MoreJobs = 1 Please enter a short title for industry	
1_40 26	•	JOB3_ETYPE NEW QUESTION APPLIES IF MoreJobs = 1 Please look at this card and tell me which of these best describes your employment situation in this job Employee Running a business or professional practice Partner in a business or professional practice Working for myself A subcontractor (including SC60) Doing freelance work Self-employed in some other way	(1) (2) (3) (4) (5) (6) (7)
1_40 31	÷	JOB3_DirNI NEW QUESTION APPLIES IF Director = 1 In this job, are/were NI contributions deducted at source? Yes No	(1) (2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
-	1_40 35	JOB3_WORKSTAT No longer asked Now a derived variable Code 1 if EType = 1 Else code 2	
		Pay Details	
		1995/6 QUESTIONS I_50 345 TO I_50 350 AND I_50 360 TO I_50 3 DELETED	65
	1_50 5	PAYMAIN_ANTICPAY DELETED	
I_50 15	I_50 15	PAYMAIN_PAYAMT REWORDED What was your LAST take home pay, (what will be your take home p including overtime, bonus, commission or tips?	ay)
	I_50 30	PAYMAIN_PAYEF DELETED	
[_50 35	1_50 35	PAYMAIN_TAX REWORDED How much was deducted from your wage/salary for income tax under	PAYE?
	1_50 60	NINOf DELETED	
I_50 65	1_50 65	PAYMAIN_NINO REWORDED How much was deducted as a National Insurance Contribution?	
I_50 110	1_50 110	DEDUCTS, DEDUCTS1 through to DEDUCTS8 CODING CHANGE Superannuation? AVCs (additional Voluntary Contributions) Union fees? Friendly Societies Sports or social clubs Repayment of loan from employer Private Medical Insurance Or any other deductions? NONE OF THESE	(1) (2) (3) (4) (5) (6) (7) (8) (9)
1_50 175	-	INSAMT NEW QUESTION APPLIES IF Deducts = 7 How much was deducted for private medical insurance?	
	I_50 230	PayOTf DELETED	

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
1_50 235	-	PayOtf NEW QUESTION APPLIES IF PayDat in last 12 months Do you ever do any work which you would regard as paid or unpaid overti Yes No	me [?] (1) (2)
I_50 240	-	HourUs NEW QUESTION APPLIES IF in work or worked in last 12 months How many hours a week do/did you USUALLY work in a week? Please EXCLUDE meal breaks (and overtime)	
I_50 245	I_50 245	HourOt REWORDED How many hours PAID overtime do/did you USUALLY work per week?	
I_50 250	-	HourNopa NEW QUESTION APPLIES IF PayOTf = 1 How many hours unpaid overtime do/did you usually work in a week?	
	I_50 240	SSSMPay DELETED	
	I_50 245	SSPay DELETED	
I_50 255	-	PAYINCO PAYINC3 NEW QUESTION APPLIES IF PayDat in last 12 months Did your take home pay include any of the items shown on card P1? Statutory Sick Pay Statutory Maternity Pay Mileage allowance/fixed allowance for motoring Motoring expenses refund None of these	(1) (2) (3) (4) (5)
1_50 260	-	MileAll NEW QUESTION APPLIES IF Worked in last 3 months AND PayInc = 3 How much mileage or fixed motoring allowance was included?	
1_50 270	-	MotEAmt NEW QUESTION APPLIES IF Worked in last 3 months AND Paylnc = 4 How much was included for motoring expenses including vehicle purchase vehicle insurance?	and
		Self-Employment Income	
		1996/7 QUESTIONS 1_60 50 TO I_60 160 DELETED	
1_60 10	-	JobBus NEW QUESTION APPLIES IF Workstat = 2 Can I just check, do you think of yourself as having a job or a business? Job Business Neither	(1) (2) (3)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_60 I5	-	BusAccts NEW QUESTION APPLIES IF Workstat = 2 In this job/business are/were annual business accounts prepared for the Inl. Revenue for tax purposes?	and
		Yes No Not yet but will be	(1) (2) (3)
I_60 20	-	Sole NEW QUESTION APPLIES 1 Workstat = 2 Are/were you working on your own account or are /were you in partnersh someone else? Own account (sole owner) In partnership	(1) (2)
1_60 30	-	Sel NEW QUESTION APPLIES IF BusAccts = 1 What is the most recent period for which accounts have been prepared for Inland Revenue? ENTER BEGINNING OF PERIOD	· the
1_60 35	-	Se2 NEW QUESTION APPLIES IF BusAccts = 1 ENTER END OF PERIOD	
I_60 40	-	SeWeeks NEW QUESTION APPLIES IF BusAccts = ! May I just check, how many weeks does this cover?	
I_60 45	-	ProfDocs NEW QUESTION APPLIES IF BusAccts = 1 What was (your share of) the profit or loss figure shown on these account this period? It would be helpful if you could refer to a document, such as accounts, or the Notice of Tax Assessment from the Inland Revenue Notice of Tax Assessment (form no 300 - CODA) Annual Accounts (incl summary) Some other document (describe in a note) No document consulted	
1_60 50	-	Profit1 NEW QUESTION APPLIES IF BusAccts = 1 ENTER THE £PROFIT/LOSS AMOUNT (from Notice of Assessment en income figure at top of form/from Accounts, enter adjusted profit/loss - 11 shown enter net figure)	
I_60 55	-	Profit2 NEW QUESTION APPLIES IF Profit > 0 DID THE ANSWER IN THE PREVIOUS QUESTION REFER TO PRO OR LOSS? Profit Loss	0FIT (1) (2)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_60 65	-	ProfTax NEW QUESTION APPLIES IF Profit2 = 1 Can I just check, is that the figure before deduction of income tax? Yes (before tax) No (after tax)	(1) (2)
I_60 70	-	ProfNI NEW QUESTION APPLIES IF Profit1 = 0 OR ProfTax = 2 And is the figure before or after you paid any lump sum (Class4) National Insurance payments based on taxable profits? Before After Not applicable (no lump sum NI)	(1) (2) (3)
I_60 75	-	PrBefore NEW QUESTION APPLIES IF Profit1 = 0 OR PrfoTax = 2 What was (your share of) the profit BEFORE tax (and lump sum deductions)	ons) ⁹
1_60 85	-	WhyNoPro NEW QUESTION APPLIES IF Profit! = DK Why was respondent unable to give a profit or loss figure? Documents with accountant/Inland Revenue Other reason	(1) (2)
I_60 90	-	WorkAcc NEW QUESTION APPLIES IF BusAccts = 1 or 3 Do you have separate bank or building society accounts for your work an private finances? Yes No	(1) (2)
1_60 95	-	OwnSum NEW QUESTION APPLIES IF WorkAcc = 1 Do you draw money from your work account for any non-business purpo such as any of the things shown on this card? Yes No	ses, (1) (2)
I_60 100	-	OwnAmt NEW QUESTION APPLIES IF OwnSum = ! Thinking of the last 12 months, on average how much did you take EAC MONTH for these non-business purposes?	Н
I_60 110	-	OwnOther NEW QUESTION APPLIES IF OwnSum = I Apart from drawings from the bank/building society, do you receive any income from this job/business, for personal use? Yes No	other (1) (2)
1_60 115	-	OwnOtAmt NEW QUESTION APPLIES IF OwnOther = 1 On average, how much is that each month?	

Questionnaire	Changes
---------------	---------

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change
I_60 125	-	SEInc NEW QUESTION APPLIES IF BusAccts = 2 or DK OR WorkAcc = 2 or DK OR OwnSum = 2 or DK Now I'd like to ask some questions about your income from your job/business that is, after paying for any materials, equipment or goods that you use(d) in your work On average, what is your WEEKLY or MONTHLY income from this job/business over the last 12 months?
1_60 130	-	SEIncWM NEW QUESTION APPLIES IF SEInc > 0 Was that weekly or monthly income? Weekly income Monthly (2
I_60 140	-	CheckTax NEW QUESTION APPLIES IF SEInc > 0 May I just check, was either income tax, or your regular National Insurance contribution DEDUCTED AT SOURCE? income tax deducted regular NI deducted no, neither deducted (3)
I_60 145	-	TaxDAmt NEW QUESTION APPLIES IF CheckTax = 1 How much income tax was deducted last time?
I_60 150	-	PercTaxD NEW QUESTION APPLIES IF TaxDAmt > 0 How long did this cover?
I_60 160	-	NIDAmt NEW QUESTION APPLIES IF CheckTax = 2 How much National Insurance was deducted last time?
I_60 165	-	PercNID NEW QUESTION APPLIES IF NIDAmt > 0 How long did this cover?
I_60 175	-	ChkIncom NEW QUESTION APPLIES IF CheckTax = 1 or 2 May I check, is your average income of £ before or after income tax/regular it was deducted? Before After (
I_60 180	-	SeNIReg NEW QUESTION APPLIES IF Self-employed and CheckTax = 1,3 or empty AND male 16-65 or female 16-60 Do you pay a regular National Insurance contribution in connection with this Job? (KNOWN AS "CLASS 2") Yes No (

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_60 1 8 5	-	SeNIRAmt NEW QUESTION APPLIES IF SeNIReg = 1 How much was your last National Insurance payment?	
I_60 190	-	PercSeNI NEW QUESTION APPLIES IF SeNIReg > 0 How long did this cover?	
1_60 200	-	SeTax NEW QUESTION APPLIES TO ALL SELF-EMPLOYED (Apart from tax deducted at source) Have you made any (other) income tax payments relating to this job/business in the last 12 months? Yes No	(1) (2)
1_60 205	-	SeTaxAmt NEW QUESTION APPLIES IF SeTax = 1 How much did you pay altogether in the last 12 months?	
I_60 215	•	SeNInc NEW QUESTION APPLIES IF SeTaxAmt = RESPONSE Does that figure include a LUMP SUM (Class 4) National Insurance contribution bases on taxable profits? Yes No	(1) (2)
1_60 220	-	SeNIAmt NEW QUESTION APPLIES IF SeNImc = 1 How much was the National Insurance lump sum payment?	
I_60 230	-	SeNILump NEW QUESTION APPLIES IF SeNImc = 2 OR DK In the last 12 months have you paid any lump sum NI contributions based of taxable profits? Yes No	on (1) (2)
I_60 235	-	SeNiLAmt NEW QUESTION APPLIES IF SeNILump = 1 What was your total lump sum payment in the last 12 months?	
	I_70 10	Pay from subsidiary job as employee PAY2_ANTICPAY DELETED	
	I_140 5	State Benefits(1) FiltPr DELETED	

Questionnaire	Changes
---------------	---------

1996-97 question	1995-96 question		
number	number	1996-97 Questionnaire text or description of change	
1_140 10	1_140 10	BENR1_BNCARD00 through to BENR1_BNCARD09 REWORDED AND CODING CHANGE Looking at card R1, are you at present receiving any of these state benefits your own right that is, where you are the named recipient? Child benefit One parent family benefit Guardian's Allowance Invalid Care Allowance Retirement pension (National Insurance) Old person's pension Widow's pension or allowance (NI) War disablement pension Severe disablement allowance Disability Working allowance None of these	(1) (2) (3) (4) (5) (6) (7) (8) (9) (10) (11)
		State Benefits (2)	
	l_150 5	FiltP DELETED	
I_150 10	[_150 10	BENR2_BNCARDR0 through to BENR2_BNCARDR2 REWORDED AND CODING CHANGE And looking at card R2, are you at present receiving any of these state bereither in your own name, or on behalf of someone else in this household? CARE COMPONENT of Disability Living Allowance MOBILITY COMPONENT of Disability Living Allowance Attendance allowance None of these	(1) (2) (3) (4)
		State Benefits(3)	
	[_160 10	Filt12 DELETED	
I_160 15	1_160 15	BENSI_BNCARD00 through to BENSI_BNCARD05 REWORDED AND CODING CHANGE Now looking at card S1, have you received IN THE LAST 12 MONTHS since, any of these benefits in your own right that is, where you are the named recipient? N1 Unemployment Benefit Income Support Family Credit Incapacity Benefit Statutory Sick Pay (from employer) Industrial injury disablement benefit None of these	(1) (2) (3) (4) (5) (6) (7)
		State benefits(4)	
	1_170 5	FiltMt DELETED	

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
1_170 10	I_170 IO	BENS2_BNCARDS, BENS2_BNCARDS1 REWORDED AND CODING CHANGE In the last 12 months, have you received either of the things shown on carrin your own right? Maternity Allowance Statutory maternity pay from your employer or former employer Neither of these	d S2, (1) (2) (3)
I_170 15	-	BENS2_NUMWKS NEW QUESTION APPLIES IF I_170 10 = 1 For how many weeks in the last 12 months since (date) did you receive the benefit?	ıs
I_170 20	-	BENS2_AMTBENX NEW QUESTION APPLIES IF I_170 10 = 1 What was the amount of benefit you last received?	
I_170 25	-	BENS2_PERBENX NEW QUESTION APPLIES IF I_170 10 = 1 What period does this cover?	
I_170 30	•	BENS2_RCPTNOW NEW QUESTION APPLIES IF 1_170 10 = 1 Are you receiving this benefit at present? Yes No	(1) (2)
I_170 40	-	BENS2_NUMWKS1 NEW QUESTION APPLIES IF I_170 10 = 2 For how many weeks in the last 12 months since (date) did you receive the benefit?	ıs
1_170 45	-	BENS2_RCPTNOW1 NEW QUESTION APPLIES IF I_170 10 = 2 Are you receiving this benefit at present? Yes No	(1) (2)
		State Benefits(5)	
	I_180 5	FiltO DELETED	
I_180 10	1_180 10	BENTI_BNCARD00 through to BENTI_BNCARD04 REWORDED AND CODING CHANGE In the 12 months since (date), have you received any of the things shown card T1, in your own right? Family Credit paid as a lump sum	on (1)
		Grant from Social Fund for Funeral Expenses Grant from Social Fund for Maternity Expenses	(2)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_180 15	-	BENT1_FamLAmt NEW QUESTION APPLIES IF BnCardTl = 1 What was the total Family Credit lump sum payment you received?	
I_180 25	-	BENT1_FunerAmt NEW QUESTION APPLIES IF BnCardT1 = 2 What was the total grant you received for funeral expenses?	
I_180 35	-	BENT1_MatGrAmt NEW QUESTION APPLIES IF BnCardT1 = 3 What was the total grant received for maternity expenses?	
I_180 45	-	BENT1_CCLAmt NEW QUESTION APPLIES IF BnCardT1 = 4 What was the total Community Care grant you received?	
I_180 60	-	BENT1_NUMWKS NEW QUESTION APPLIES IF BnCardT1 = 5 For how many weeks in the last 12 months since (date) did you receive the benefit?	hıs
I_180 65	-	BENT1_AMTBENX NEW QUESTION APPLIES IF BnCard1T1 = 5 What was the amount of benefit you last received?	
I_180 70	-	BENT1_PERBENX NEW QUESTION APPLIES IF BnCardT1 = 5 What period does this cover?	
1_180 75	-	BENTI_RCPTNOW NEW QUESTION APPLIES IF BnCardT1 = 5 Are you receiving this benefit at present? Yes No	(1) (2)
		Income from pensions and trusts	
I_200 5	1_200 5	Pensinc CODING CHANGE Are you at present receiving an income from any of these sources? SHOW PROMPT CARD UI Yes No	(1) (2)
I_200 20	-	PensType NEW QUESTION APPLIES IF PensInc = 1 Which type of pension is this? An employee pension from a previous employer Pension from employed of a deceased spouse or relative A private personal pension An annuity, home income plan, equity release plan A pension as a member of a trade Union or friendly society A payment from a trust or covenant	(1) (2) (3) (4) (5) (6)

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
I_200 20	1_200 20	IncAmt REWORDED How much did you receive last time?	
I_200 65	1_200 65	EmpDedAm REWORDED How much was/were the deductions?	
	1_200 75	ÉmpDeds DELETED	
	I_200 8 5	EmpMtOne DELETED	
1_200 90	-	PENSINC_Anymore NEW QUESTION APPLIES IF PensInc = 1 Do you have any other income from any of the sources on card U1? Yes No	(1) (2)
		1995/6 QUESTIONS 1_200 90 TO 1_200 120 DELETED	
		Regular allowances	
I_230 15	-	Alltype CODING CHANGE What allowances have you received? REGULAR allowance from a member of your household who is TEMPORARILY absent Maintenance or separation allowance Any other REGULAR allowance from an INDIVIDUAL who is N member of this household Allowance for foster child Allowance for adopted child Other regular allowance from an organisation	(1) (2) OT a (3) (4) (5) (6)
I_230 20	-	ALLOW_PerWho NEW QUESTION APPLIES IF AllType = 1 Which member of your household gave you the money? Absent spouse Other absent spender	(1) (2)
1_230 25	-	IndPer NEW QUESTION APPLIES IF AllType = 3 Who gave you this allowance? A spouse who is not a household member Another relative who is not a household member Some other individual	(1) (2) (3)

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change
1_230 30	-	MntWho NEW QUESTION APPLIES IF AllType = 2 Is this maintenance or separation allowance for yourself only (1) child/children only (2) or yourself and child/children (3)
1_230 35	I_230 20	AllRecam REWORDED How much did you receive last time?
I_230 40	I_230 25	Perc584 NEW QUESTION APPLIES IF AllRecam = valid value What period does this cover?
I_230 50	1_230 35	AllPres NEW QUESTION APPLIES IF AllRecam = valid value Are you receiving this allowance at present? Yes No (1) (2)
I_230 55	-	ALLOW_Anymore NEW QUESTION APPLIES IF Filter = 1 In the 12 months since (date) have you received any more of these allowances? Yes (1) No (2) 1995/6 QUESTIONS I_240 5 TO I_250 35 DELETED
I_290 31	-	TESSAs TesNam NEW QUESTION APPLIES IF TesAmt = DK May I check, with which bank or building society do you hold your TESSA?
1_300 26	-	Building society accounts BIdNam NEW QUESTION APPLIES IF BIdAmt = DK What is/are the names of the accounts you hold?
I_310 26	-	High street bank accounts BkNam NEW QUESTION APPLIES IF BkAmt = DK What is/are the names of the accounts you hold?

Family Expenditure Survey

Questionnaire Changes

1996-97 question number	1995-96 question number	1996-97 Questionnaire text or description of change	
		Gilts, unit trusts, PEPS, other shares and securities	
1_350 5	1_350 5	Filter CODING CHANGE Do you have now, or have you had in the last 12 months since (date),	income
		from any of these investments? Government gift-edged stock/war loan	(1)
		Unit and investment trusts	(2)
		Personal Equity Plans	(3)
		Other stocks, shares and bonds including National Savings	• /
		Investment	(4)
		NONE OF THESE	(5)
I_350 40	-	OthNam NEW QUESTION APPLIES IF OthAmt = DK What is/are the names of the accounts you hold?	